

The Impact of Policy Interventions on Systemic Risk across Banks

Simona Nistor¹ · Steven Ongena²

Received: 8 January 2021 / Revised: 26 January 2023 / Accepted: 31 January 2023 / Published online: 22 February 2023 © The Author(s) 2023

Abstract

What is the impact of policy interventions on the systemic risk of banks? To answer this question, we analyze a comprehensive sample that combines an original set of bank-specific bailout events with the balance sheets of key affected and nonaffected European banks between 2005 and 2014. We find a positive and significant association of *guarantees* with systemic risk that is somewhat weaker in the long run when the regulator appoints members to the supervisory board. The short run association between *recapitalizations* and systemic risk is also positive for large and less capitalized banks, while in the long run, recapitalizations are linked with reduced systemic importance, especially for less profitable banks and in cases when the regulator limits management pay. *Liquidity injections* are positively linked with systemic risk, but the long-run effect is mitigated for small or better capitalized banks. In the short run, injecting liquidity is associated with reduced systemic risk when the regulator imposes restrictions on supervisory board composition or on management pay or capital payouts.

Keywords Systemic risk · Policy interventions · Risk strategies · Regulatory restrictions

JEL Classification $E58 \cdot G01 \cdot G21 \cdot G28 \cdot H81$

1 Introduction

The recent financial crisis led to coordinated efforts by governments and central banks to avoid a major systemic meltdown. Public interventions such as debt guarantees, capital injections, state loans, acquisitions of impaired assets, and/or nationalizations were

Steven Ongena steven.ongena@bf.uzh.ch
 Simona Nistor simona.mutu@econ.ubbcluj.ro

¹ Faculty of Economics and Business Administration, Babeş-Bolyai University of Cluj-Napoca, Teodor Mihali Street, Nr. 58-60, 400591 Cluj-Napoca, Romania

² University of Zürich, Swiss Finance Institute, KU Leuven, NTNU Business School and CEPR, Plattenstrasse 14, 8032 Zürich, Switzerland

implemented on an unprecedented scale by most countries.¹ At the European Union (EU) level, policy actions adopted by member states immediately after the Lehman Brothers collapse were coordinated in a massive bailout of financial institutions that was estimated to have amounted to 3.65 trillion euro (European Commission 2009).² Since then, several additional financial support programs have been set up, especially after the European sovereign debt crisis and the Greek bailout in 2010.³ These types of emergency assistance programs play an important role in restoring public confidence in the banking sector. However, how effective are these different tools in controlling systemic risk, and how heterogeneous is their impact across banks' risk strategies? Given the current COVID-19 pandemic and its potential dire economic and financial consequences for the banking sector, we think this is a highly relevant policy question.

Our paper contributes to this literature by investigating the interplay between systemic risk, policy interventions, regulatory restrictions, and banks' risk strategies. Our study analyses 83 key global and local financial institutions from 22 European countries during the 2005–2014 period (see Online Appendix 1). First, we are interested in assessing the immediate effects of policy interventions on systemic risk. Second, we examine their long-run implications. While in the short run regulators aim to restore confidence and alleviate the spread of contagion through interventions, over a longer period, for the bailed-out banks, the too-big-to-fail protection may be traded off with the costs and restrictions associated with bailouts. Moreover, we explore how regulatory restrictions and banks' risk strategies exacerbate or mitigate the relation between rescue actions and systemic risk. The main research questions that we aim to answer therefore are as follows: *What are the immediate effects of policy interventions affect systemic risk in the long run*?

Clearly, these elements are interdependent. For example, it is more likely that the government provides bailouts to systemically important banks, which due to their size, higher risk, complexity, and/or interconnectedness can have a sizeable adverse impact on financial stability in case of their failure (Berger et al. 2020; Gerhardt and Vander Vennet 2017; Kick et al. 2016). In our empirical analysis, we aim to methodically account for this interdependence using a Heckman selection procedure. The systemic risk indicators are estimated based on the loss generated by the reduction in the banks' market capitalization under extreme events. We use the *Marginal Expected Shortfall (MES)* of Acharya et al. (2017), a measure that captures systemic risk realizations, as

¹ We use the following terms as synonyms: policy interventions, public interventions, bailouts, financial assistance programs, and emergency rescue measures.

² Broadly, bailouts reflect the financial support extended to a company or a country facing financial difficulties and may come in various forms. In our paper, bailouts represent the financial assistance provided by governments and central banks to financial institutions and take the form of guarantees, recapitalizations, and liquidity injections.

³ In May 2010, the European Union member states set up the European Financial Stability Facility (EFSF) that has a maximum lending capacity of 440 billion euro and the European Financial Stabilization Mechanism (EFSM) with a lending capacity of 60 billion euro. Funds are raised through borrowing from financial markets and are guaranteed by the European Commission with the European Union budget as collateral. On September 27th, 2012, these two funds were replaced by the European Stability Mechanism (ESM) with the aim of providing instant financial assistance to Eurozone members with a limit of 500 billion euro (ESM Annual Report 2013).

well as forward-looking risk. The negative spillovers from each bank to the system are defined as their contribution to systemic risk or their systemic importance.⁴

Although during recent years several financial assistance programs have been set up to combat the Eurozone crisis,⁵ we focus on the emergency rescue measures most used by European member states during the global financial crisis (GFC) and the European sovereign debt crisis (ESDC). These public interventions, supported by governments or central banks at the national level and agreed to by the European Commission, consisted of instruments designed to limit systemic risk and the spread of contagion during the financial crises in Europe. Our interest resides in assessing the effectiveness of these most flexible policy interventions in controlling systemic risk. To perform this evaluation, we use a unique dataset of bank-level interventions by national authorities (state guarantees, recapitalizations and liquidity injections) collected from banks' annual reports, financial statements, websites and the State Aid Register of the European Commission (see Online Appendix 2 and Online Appendix 3).

Our empirical findings establish that bailouts are associated with different evolutions of systemic risk in the short versus long run. We find evidence that in the short run, banks that receive policy interventions are linked with an enhanced contribution to systemic risk and that the effect is strongly significant for state guarantees. We argue that the positive association with systemic contribution might be attributed to the realization of risk captured by the MES measure. Interventions are usually implemented after risk is realized. As our dependent variable reflects both systemic risk realizations and forward-looking risk, the valuation of banks by investors is likely to be reduced before intervention and to not recover for some time after. The delayed success of the interventions may also be partly because European states were not always credible providers of financial assistance during the period analyzed. In the long run, liquidity injections – by providing only temporary relief – are associated with enhanced systemic risk, while recapitalizations somewhat are linked with reduced systemic importance. The findings suggest that banks injected with liquidity remained weak and investors penalized their stock returns, which is reflected in a higher systemic contribution. In turn, recapitalizations fix the problem created by risk realizations. We further provide empirical evidence that the link between policy interventions and systemic risk varies when restrictions such as supervisory board intrusions, management pay limitations, and capital payout bans are imposed and across banks with different risk strategies related to their size, leverage, and profitability.

Based on our estimates, important policy conclusions can be drawn. First, emergency policy interventions should be adequately implemented because their association with systemic risk could be different in the long versus short run. Second, their effectiveness can be significantly influenced both by the restrictions imposed by the regulator throughout the duration of the financial assistance and by the banks' risk strategies.

The remainder of the paper is organized as follows. Section 2 provides the literature review. Section 3 describes the methodology. Section 4 introduces the sample and the data. Section 5 presents the empirical results. Finally, Section 6 concludes.

⁴ We use the following terms as synonyms: contribution to systemic risk, systemic contribution and (at times) systemic importance.

⁵ The EFSF, EFSM and ESM funds, the EU's Balance of Payments program (BoP), and bilateral loans from the International Monetary Fund (IMF), World Bank, European Investment Bank (EIB) and European Bank for Reconstruction and Development (EBRD).

2 Literature review

2.1 General

This paper connects several strands of literature related to studies assessing risk realizations and research focusing on forward-looking risk. First, our paper is linked to research on policy interventions and risk realizations. A number of studies analyze how regulatory policies can help control systemic risk. Considering several international financial crises, Weiß et al. (2014a), for example, find that global systemic risk is significantly influenced by the characteristics of regulatory regimes, explicit deposit insurance schemes and country-wide (macro-) prudential policy interventions (e.g., Karamysheva and Seregina 2020). Recapitalizations can reduce the systemic contribution of banks (López-Espinosa et al. 2012), while liquidity injections can temper the risk incentives of insolvent institutions (Cordella and Yeyati 2003). Buch et al. (2019) show that European banks that received state aid during the crisis are associated with increased systemic importance. However, the U.S. Troubled Assets Relief Program (TARP), based on injections of preferred equity, significantly reduced contributions to systemic risk, particularly for larger and safer banks (Berger et al. 2020). Anginer et al. (2014b) point to the stabilizing role played by deposit insurance arrangements during stress periods but also criticize their destabilizing role in normal times.

Second, our research is related to an extensive literature exploring the forwardlooking nature of risk. Numerous papers investigate the moral hazard embedded in deposit insurance schemes (Demirgüç-Kunt and Detragiache 2002; Demirgüç-Kunt and Huizinga 2004; Gropp et al. 2014), recapitalizations (Kane 1995), or liquidity injections (Acharya and Yorulmazer 2007; Farhi and Tirole 2012). Financial help in general may deteriorate the liquidity situation of banks when regulators cannot and/or do not distinguish between illiquid and insolvent financial institutions (Freixas et al. 1999; Repullo 2005). Moreover, rescue packages provided to large banks may incentivize them to engage in highly risky operations (Mishkin 2006), invest in illiquid assets (Cao and Illing 2008) and take on excessive credit risk (Gropp et al. 2014). Additionally, state guarantees provided to efficient banks can enhance their exposure to systemic events (Myerson 2012). On the other hand, the adverse impact of a deposit insurance scheme on systemic risk may diminish if banks hold higher levels of Tier 1 capital (Bostandzic et al. 2014). Homar (2016) highlights the importance of the amount of interventions, presenting empirical evidence that banks that receive large enough capital injections boost the supply of credit, access supplementary funding and improve their balance sheets. Similarly, Giannetti and Simonov (2013) show that a reasonable level of capital injections helped banks increase lending and stimulate investments during the Japanese banking crisis of the 1990s. In the long run, bailouts can increase investors' expectations of future bailouts (Bayazitova and Shivdasani 2012), especially when banks are considered too big to fail, too interconnected to fail and/or too many to fail (Acharya and Yorulmazer 2007; Brown and Ding 2011). This could generate enormous costs for shareholders and taxpayers, impede a recovery, and distort competition. Rescued banks may obtain competitive advantages that increase their market power (Berger and Roman 2015), while sound banks that do not receive intervention may increase their loan rates and reduce depositor risk premiums (Koetter and Noth 2016).

2.2 Contributions

We contribute to the extant literature in several ways. First, our foremost contribution is to assess to what extent policy interventions implemented by governments affect systemic risk. Although the isolated impact of the bailout mechanism has been addressed in several theoretical and empirical studies, in comparison with the approaches in other studies, our specifications include several types of interventions with details regarding their volume and their associated restrictions. A large spectrum of the most important policy interventions made by European member states and implemented at the bank level during 2008–2014 is examined: state guarantees, recapitalizations (capital injections) and liquidity injections. The dataset is collected manually from banks' annual reports, financial statements, websites, and the State Aid Register of the European Commission. Our empirical approach is also different. First, we are interested in investigating the short-run response of systemic risk to policy interventions; in addition, we examine the long-run implications of this relation. This distinction is particularly important for European countries, as governments were not always perceived as trustworthy providers of financial support, which might lead to a delayed effect of interventions.

Second, we contribute to the literature on the restrictions associated with bailouts. In the long run, a tradeoff between too-big-to-fail (TBTF) protection and the regulatory burdens associated with bailouts might come into play. We employ a hand-collected dataset of regulatory restrictions and assess their impact on the relation between policy interventions and systemic risk. Specifically, we consider the following set of restrictions imposed by regulators on the banks receiving intervention during the duration of bailouts: supervisory board intrusions, management pay limitations, and capital payout bans.

Third, we add to the literature by examining the interaction between policy interventions and banks' risk strategies. We explore how banks' risk postures related to size, leverage, and profitability exacerbate or mitigate the relation between rescue actions and systemic risk. Especially in the long run, banks are more likely to manifest moral hazard behavior by adjusting their risk strategies.

3 Methodology

This section presents the regression specifications used to analyze the impact of the emergency policy interventions made by European national supervisory authorities during the global financial crisis and the European sovereign debt crisis on systemic risk. First, we assess the effects of policy interventions on systemic risk in the short run and in the long run. Second, we explore how banks' regulatory restrictions and risk strategies exacerbate or mitigate the relation between rescue actions and systemic risk. We employ the Heckman selection approach and control for a variety of bank, market, and macro characteristics, using a sample of 83 publicly listed European banks, with data spanning from 2005 to 2014.

3.1 Identification

To account for the selection of banks into treatment (government intervention) on their unobservable systemic importance, we employ the two-stage procedure of Heckman (1979). This approach is in line with the previous literature (Berger et al. 2020; Gerhardt

and Vander Vennet 2017; Kick et al. 2016). In the first stage, we use a *Probit* model to estimate the probability of a bank receiving policy interventions, as represented by the equation:

$$Ln(P_{Intervened \ ij,t}) = \beta_0 + \Omega \times Identifying \ restrictions_{ij,t-1} + \Phi \times Bank \ controls_{ij,t-1} \Psi$$

$$\times Market \& \ Macro \ controls_{i,t-1} + v_t + \varepsilon_{ij,t}$$
(1)

where $P_{Intervened ij,t}$ represents the probability that bank i receives (one or more) policy interventions in quarter t, *Identifying restrictions*_{ij,t-1} is a set of exclusion restrictions that explain the decision of government to provide financial assistance to bank i, and *Bank controls*_{ij,t-1} represents differences in risk profiles among banks (size, leverage, credit risk, liquidity and profitability). To account for heterogeneity among different banking systems and economies, we include banking market controls (*Market controls*_{j,t-1}) and macroeconomic controls (*Macro controls*_{j,t-1}), which are at the country level. The specifications include year fixed effects (v_t) to control for unobserved heterogeneity. $\varepsilon_{ij,t}$ is an *iid* error term specific to bank i from country j in quarter t. The explanatory variables are lagged by one period. Variables are winsorized at the 1st and 99th percentiles. The results are corrected for heteroskedasticity and correlation using bank-level clustered standard errors.⁶ The empirical specification is run for the full sample of 83 banks, and the period accounts for 40 quarters during 2005–2014.

The Probit model includes several identifying restrictions that are excluded from the 2nd-stage OLS specifications, i.e., political stability, vote share of nongovernment parties, legislative and executive elections dummy, and prevalence. The political stability index measures the stability of political institutions, with higher values indicating greater political stability. The vote share of nongovernment parties reflects the vote share of parties other than the governing parties. The legislative and executive elections dummy is a variable that takes the value 1 if a legislative or executive election took place in the quarter before bailouts were provided. Prevalence is the index developed by Braun and Raddatz (2010) and measures the prevalence of connectedness through the ratio of actual to expected number of political connections (i.e., in essence the number of cases within a country where a former politician later sits on a bank's board).⁷ The political stability index is from the World Governance Indicators database, the vote share of nongovernment parties and the legislative and executive elections dummy are from the Database of Political Institutions of Cruz et al. (2016), and the prevalence index is from Braun and Raddatz (2010). The choice of the exclusion restrictions follows the literature suggesting that political institutions and electoral cycles influence banks' probability of receiving financial assistance (Brown and Dinc 2005; Liu and Ngo 2014; Behn et al. 2017; Berger et al. 2020).

An important condition that the identifying covariates must satisfy is that they should directly affect the probability of intervention (i.e., one or more of the exclusion variables) but not be correlated with systemic risk. To verify the validity of the exclusion variables,

 $^{^{6}}$ In different exercises, we use standard errors clustered at country level, bank×year level, and bank×quarter level and two-way clustering by bank and quarter. The (unreported) results confirm that the significance of the main regressors is unaffected.

⁷ To alleviate the concern that countries with more connections may be countries with fewer people, Braun and Raddatz (2010) derive the probability that a former politician later sits on a bank's board by assuming that the connections are selected randomly with replacement from a common pool. This probability gives the expected number of political connections within a country.

we assess their explanatory power for the likelihood of bailouts and their orthogonality with systemic risk. Following Dam and Koetter (2012), we employ two econometric tests. First, using a weak identification test, i.e., a pseudo Kleibergen–Paap F test, we check whether the set of identifying restrictions have sufficient explanatory power for the likelihood of intervention in the 1st-stage equation. Second, we examine the exogeneity of the excluded covariates, i.e., whether they are uncorrelated with systemic risk, employing a pseudo Hansen J test.⁸

3.2 Baseline model: short-term and long-term effects

In the second stage of the Heckman (1979) procedure, we examine the impact of emergency policy interventions on systemic risk using an *OLS fixed effects* model for the sample restricted to the 30 rescued banks. To control for sample selection bias, the inverse Mills ratio generated by the Probit model in Eq. (1) is included. The following baseline model specification is run for the restricted sample of banks that received bailouts:

$$SystemicRisk_{ij,t} = \beta_0 + \beta_1 \times Policy interventions_{ij, event window} + \beta_2 \times IMR_{ij,t-1} + \Phi \times Bank \ controls_{ij,t-1} + \Psi \times Market \& Macro \ controls_{j,t-1} + \varphi_i + v_t + \mu_{jt} + \varepsilon_{ij,t}$$
(2)

The dependent variable is represented by the contribution of bank i from country j to systemic risk in quarter t. The data represent the values of bank-level systemic risk indicators estimated on a weekly basis using the MES methodology proposed by Acharya et al. (2017).⁹ The MES assesses the marginal contribution of a bank to the total capital shortfall of the system. Using this methodology, we estimate the average return on bank i's market capitalization in the weeks in which the total market capitalization of the system experiences the worst 1% of its outcomes (i.e., the expected loss of bank i's market equity returns conditional on the system's market equity returns exceeding the Value at Risk limits). We start by determining the conditional Expected Shortfall (ES) of the system's returns as follows: $ES_t^{sys} = E\left[R_{Market Equity,t}^{sys} | R_{Market Equity,t}^{sys} | S_{Market Equity,t}^{sys} \le VaR_t^{sys}\right]$, where $R_{Market Equity,t}^{sys}$ is the return of the system's market capitalization and VaR_t^{sys} is the Value at Risk indicator that expresses the maximum possible loss that the system can register for a given confidence level α (i.e., 1%) over a specific period of time.¹⁰ To estimate the values of MES, we use a multivariate GARCH-DCC specification, a refinement proposed by Brownlees and Engle (2017) that accounts for time-varying volatility and correlation. Online Appendix 4.1 provides a detailed description of the estimations. To merge the quarterly balance sheet and macroeconomic variables, we transform the

⁸ We use the term "pseudo" because we do not know what would have happened with systemic risk in the absence of the interventions.

⁹ We also employ the Conditional Value at Risk indicator of Adrian and Brunnermeier (2016) as a robustness check.

¹⁰ The theoretical research featuring the widely used Value at Risk indicator was initiated by Jorion (1997), Dowd (1998) and Saunders (1999). Technically, this loss "is found in the left tail" of the returns' distribution function of the market capitalization and involves the estimation of the next loss functions: $Prob\left(R_{Market Equily,t}^{Sys} \leq VaR_{t}^{Sys}\right) = \alpha$.

weekly values of the systemic risk indicators to quarterly frequency by summing them up for each bank within each quarter. We express the systemic risk indicators (losses) as positive numbers; hence, higher values denote greater systemic importance.¹¹

The main regressors of interest are represented by the emergency rescue measures taken by the national authorities of country j and implemented by bank i during the event window (*Policy interventions*_{ij, event window}). These take the form of state guarantees, recapitalizations, and liquidity injections. They are represented by continuous variables obtained by dividing the volume of bailouts by the size of the banks. First, we assess the *short-run* effects of policy interventions by employing an event window starting one quarter before the intervention and ending one quarter after the intervention. Interventions take the same value within this event window (i.e., from t-1 to t+1). However, for some banks' rescue packages, a longer period might be required for their effects to be fully realized. Therefore, in the next step, we analyze the *long-run* impact of bailouts on systemic risk by employing an event window approach starting one quarter before the intervention is implemented by bank i and ending the quarter in which the bailout is paid back (i.e., through t-1 to t+n). Under this framework, the volume of bailouts as a share of total assets takes the same value during the event window.¹² A negative β_1 coefficient is associated with a decrease in systemic risk contribution for the rescued banks after they receive the assistance package from government.¹³

 $IMR_{ij,t-1}$ represents the inverse Mills ratio generated by the *Probit model* in Eq. (1). The descriptions of the other regressors are similar to those for Eq. (1). The specifications include bank fixed effects (φ_i), year fixed effects (v_i), and/or country×year fixed effects (μ_{ij}) to control for unobserved heterogeneity and shocks that affected our sample countries, such as regulatory changes.¹⁴ $\varepsilon_{ij,t}$ is an *iid* error term specific to bank i from country j in quarter t. The explanatory variables are lagged by one period. Several alternative models that include other proxies for bank-level risk profiles are estimated to test the robustness of the results. Variables are winsorized at the 1st and 99th percentiles. The results are corrected for heteroskedasticity and correlation using bank-level clustered standard errors,¹⁵ especially in the context that the dependent variables are preestimated.

A limitation of our analysis is that causality cannot be established with certainty due to the lack of a natural experimental setting within our framework. To address this issue, we provide alternative explanations to the maximum extent possible.

¹¹ As Fig. 2 shows, our systemic risk measure is persistent over time. A possible way to address this issue is to include the lagged dependent variable among the regressors. However, combining the lagged dependent variable with bank fixed effects in the same regression may generate biased estimated coefficients (Nickell 1981; Flannery and Hankins 2013). Therefore, we decide to follow the common practice in the literature and omit the lagged dependent variable. In a robustness check detailed in subsection 5.1.3. Robustness, we account for the persistent nature of systemic risk by including the lag of the dependent variable among the regressors. The results remain very similar to the main findings.

¹² Alternatively, we use event windows of one year (i.e., policy interventions take the same value from one quarter before the event to four quarters after the event), two years, three years, four years, and all periods after the intervention takes place.

¹³ In the remainder of the text, the expression "systemic risk is negatively influenced" is synonymous with a decrease in systemic risk and *vice versa*.

¹⁴ For example, in 2009, the European Commission introduced a directive that required EU countries to increase their protection of retail deposits to a minimum of \notin 50,000 and then to a minimum of \notin 100,000 by the end of 2010.

 $^{^{15}}$ In different exercises, we use standard errors clustered at the country level, bank×year level, and bank×quarter level and two-way clustering by bank and quarter. The (unreported) results confirm that the significance of the main regressors is unaffected.

Policy interventions can exert an immediate effect on systemic risk, a delayed effect, or no effect at all, as depicted in Fig. 1. Moreover, the beginning of our event windows likely precedes the arrival in the market of information about the intervention. Regulators aim to fix systemic distress through financial assistance packages and usually provide them after risk is realized. Thus, declines in the equity value of rescued banks are likely during the time before intervention. An "immediate effect" would imply a prompt drop in systemic risk. However, interventions might not succeed immediately and might lead to a "delayed effect" characterized by an increase in systemic risk first and then a reduction, or measured systemic risk may increase before the intervention and be reversed slowly.



As our dependent variable incorporates pre-intervention risk realizations, interventions can be positively linked with systemic risk for a period after their implementation. In addition, European governments faced credibility concerns during the GFC and the ESDC; therefore, declines in equity value may still have occurred after the intervention. Finally, "no effect" might indicate that systemic risk increases before the intervention were fully reversed by the end of the window. The financial health of some banks might not have recovered after the intervention, leading their stock returns to continue to underperform and increasing their contribution to systemic risk even more. Nevertheless, regulators imposed restrictions on the rescued banks and imposed fees during the duration of the bailouts, which could result in even higher systemic risk.¹⁶

3.3 Further analysis. Transmission channels

3.3.1 Restrictions

Policy interventions may have increased measured systemic risk in the long or the short run due to costs associated with regulatory restrictions. In many situations, bonuses were prohibited, and dividends could be distributed only to the government. Additionally, the government occupied seats on some supervisory boards, especially in the case of recapitalizations. In addition, interventions could imply a series of commitments, such as divestments, acquisition bans, or price leadership bans, which can affect investors' expectations about the future profitability of the bank.¹⁷ For the banks in our sample, the restrictions were imposed temporarily by the regulator until the intervention was unwound.

To explore the impact of regulatory restrictions on the relation between emergency rescue actions and systemic risk, we consider the following constraints: supervisory board intrusions, management pay limitations, and capital payout bans. Intrusions on supervisory boards may ensure stricter supervision of investment and lending practices, while caps on executive compensation are likely to reduce portfolio risk (Dam and Kotter 2012). Refraining from paying dividends or from buybacks could improve banks' financial health, as retained earnings increase banks' capacity to rebuild capital buffers and promote lending. On the other hand, as capital payout limits have strong implications for shareholders, it is likely that they incentivize managers to increase portfolio risk to repay the bailout faster so that the regulator removes the restriction. Acharya and Yorulmazer (2008) show that the government's stake in the bank should be large enough to overcome risk-taking incentives.

As our MES measure captures both systemic risk realizations and forward-looking systemic risk, the effect of regulatory restrictions on the relation between interventions and systemic risk could be twofold. If shareholders perceive regulatory restrictions to be effective tools for moving portfolio risk toward optimal levels and assuring stricter monitoring of the bank, then the market valuation of more restricted rescued banks is likely to

¹⁶ In principle, we could separate effects of interventions from pre-intervention increases in systemic risk by examining the impact of interventions during three periods: one period before the official intervention date, one short period after the official intervention date, and one long period after the official intervention date. Unfortunately, this would not help us achieve separation because information about impending interventions almost surely reached markets before the official date, and the timing of information arrival varies across banks and interventions. We have chosen to begin periods of analysis one quarter before the official date so that all or almost all information about interventions is captured.

¹⁷ For a detailed analysis of the restrictions associated with European bailouts, see Panetta et al. (2009) and Berger et al. (2022).

be higher than that of less restricted rescued banks, leading to lower measured systemic importance for the former. In turn, if investors perceive the regulatory burden to be costly, then stricter restrictions may lead to underperforming stock returns and therefore higher measured systemic risk for more restricted than for less restricted bailed-out banks. This translates into a higher systemic importance for bailed-out banks when regulators impose tighter restrictions.

The impact of restrictions on the link between policy interventions and systemic risk is examined with the following specification:

$$SystemicRisk_{ij,l} = \beta_0 + \beta_1 \times Policy interventions_{ij, event window} + \beta_2 \times Policy interventions_{ij, event window} \times Restrictions_{ij,l-1} + \beta_3 \times Restrictions_{ij,l-1} + \beta_4 \times IMR_{ij,l-1} + \Phi \times Bank \ controls_{ij,l-1} + \Psi \times Market & Macro \ controls_{i,l-1} + \mu_{it} + \epsilon_{ij,l}$$

$$(3)$$

In addition to Eq. (2), we include the interaction term of policy interventions with the restrictions imposed by the regulator. The latter are captured by dummy variables that reflect the following dimensions: supervisory board intrusions, management pay limitations, and capital payout bans. The coefficient β_2 should be negative and significant if such restrictions reduce the systemic importance of banks when interventions are implemented and positive otherwise. As in the baseline specification, we use the same bank-level, market and macro controls. Additionally, we account for sample selection bias by including the inverse Mills ratio generated by the *Probit model* in Eq. (1). The strategy involves estimating the empirical models separately for each interaction of policy interventions with the restrictions using *OLS FE* for the restricted sample of rescued banks.

3.3.2 Bank risk

The interventions made by supervisory authorities may have different impacts on systemic risk across banks with varying risk profiles. While in the short run a bank's portfolio cannot change swiftly, in the long run, banks can adjust their portfolio and be more inclined to manifest moral hazard behavior. The mechanism by which increased moral hazard affects systemic risk is usually a change in the bank's risk strategy. To further explore the effects of interventions on systemic risk, we examine the impact of banks' risk strategies on the relation between emergency rescue actions and systemic risk, focusing on the following channels: size, leverage, and profitability.

Size As governments are more likely to bail out TBTF institutions, rescued banks may speculate on this behavior and increase their size to benefit from future interventions. This strategy can increase their fragility, and their contribution to systemic risk may be intensified in comparison with that of rescued banks of a smaller size. A number of empirical papers find that rescue packages provided to large banks stimulate them to focus on riskier activities and complex projects that are difficult to manage (Duchin and Sosyura 2014; Laeven et al. 2016), invest in illiquid assets (Cao and Illing 2008), or take on excessive credit risk (Gropp et al. 2014). On the other hand, bailing out a bank and confirming its TBTF status might cause the bank to downsize after the intervention due to more certainty about its TBTF status. In this case, the realization of distress, and therefore the systemic importance, may be reduced for larger rescued banks in comparison to that of smaller rescued banks.

Leverage The policy interventions implemented in the European banking sector were provided to viable financial institutions, which had to meet certain capital ratios.¹⁸ Choi (2014) theoretically shows that recapitalizing safer institutions is effective in reducing contagion spill-overs. By increasing the charter value, these banks may opt for safer investments, and their appetite for risk-taking may be reduced (Cordella and Yeyati 2003). In this way, we expect measured systemic risk to be diminished for rescued banks with a higher capitalization (i.e., above the sample median), as these banks have a greater loss-absorbing capacity than rescued banks with a lower capital ratio. At the same time, empirical evidence suggests that banks with higher excess regulatory capital may opt to take risky projects to generate returns for shareholders (Huang and Ratnovski 2009; Perotti et al. 2011). This may be especially likely to take place for banks injected with capital, as shareholders' stakes are usually diluted by the intervention. Thus, bailed-out banks with a greater capitalization may also have an enhanced contribution to systemic risk in comparison to that of rescued banks with lower capitalization.

Profitability The influence of moral hazard on systemic risk can be exerted through the profitability channel. On the one hand, a bank's higher profitability can soften the regulatory burden associated with bailouts, which leads to smaller declines in market valuation and therefore a lower systemic risk contribution. On the other hand, more profit may create conditions for banks to borrow more and take on more risk. Martynova et al. (2020), for example, show that profitable banks are more likely to make market-based investments than less profitable banks. This can lead to greater systemic risk for more profitable banks.

To assess these hypotheses, the following regression model extends the baseline specification regarding the impact of interventions on systemic risk:

$$SystemicRisk_{ij,i} = \beta_0 + \beta_1 \times Policy interventions_{ij, event window} + \beta_2 \times Policy interventions_{ij, event window} \times Bank risk_{ij,i-1} + \beta_3 \times Bank risk_{ij,i-1} + \beta_4 \times IMR_{ij,i-1} + \Phi \times Bank controls_{ij,i-1} + \Psi \times Market & Macro controls_{i,i-1} + \mu_{ii} + \epsilon_{ij,i}$$

$$(4)$$

In addition to Eq. (2), we include the interaction term of policy interventions with the bank-level risk strategy indicators. These are dummy variables that reflect the size, leverage, and profitability of the bank at t-1 (i.e., if a bank's risk is above the median risk of the sample of rescued banks one quarter before intervention).¹⁹ The coefficient β_2 should be positive and significant if risk strategies enhance the systemic importance of banks when they receive bailouts and negative otherwise. The remaining coefficients are as in Eq. (2), and the empirical strategy mimics that of the previous section. We estimate the empirical models separately for each interaction of policy interventions with the risk profile indicators.

¹⁸ The European governments provided bailouts to viable financial institutions. However, the likelihood of bailouts was higher with decreasing capitalization, as our empirical results show in Section 4. The motivation for this decision was to restore confidence in the financial system, stimulate the real economy, encourage interbank lending, or prevent competitive disadvantages from banks from the same country or from other countries.

¹⁹ To avoid multicollinearity issues, in the model with the dummy Size variable, we exclude the continuous Size variable. In the model with the Leverage dummy, we exclude the continuous Leverage variable. In the model with the Profitability dummy, we exclude the continuous ROAA variable.

This section presents the sample and the data used for estimating systemic risk measures and the variables employed in the panel regression specifications.

4.1 Sample construction

Our sample consists of 83 publicly listed European banks whose assets totaled more than 20 trillion euros at the end of 2014. They are internationally active and represent 22 European states. The interest in this portfolio is motivated by regulatory considerations, as the group includes large banks identified as G-SIBs (Global Systemically Important Banks) by Financial Supervisory Board but also small local banks that present systemic importance. Among them, 53% are included in the EBA's stress testing exercise, while 39% are included in the ECB's Single Supervisory Mechanism framework (Online Appendix 1). We focus only on banks because they are the most important financial intermediaries in Europe. The size variation is considerable within the sample, as total assets at the end of 2014 ranged from approximately 1 billion euros to approximately 2 trillion euros. The average coverage of total banking system assets of the analyzed countries is approximately 49%. We use consolidated statements to capture all cross-border business transactions of international banks.

We start with a sample of 351 active and publicly listed financial institutions from the EU28 area that are included in the Thomson Reuters Financial Datastream within the sector "Banks". We consider publicly listed banks because the methodology that we use for the estimation of systemic risk indicators is based on market data, which restricts the sample to banks listed on a stock exchange. Furthermore, due to methodology constraints imposed by the systemic risk estimation, we apply several exclusion criteria. First, we exclude banks that do not have weekly market capitalization data available in Datastream for the whole period. Second, we eliminate banks with negative equity.²⁰ Third, we include banks that have more than 75% of observations for their quarterly balance sheet data available in Worldscope (Table 1). Finally, to have a balanced sample in terms of size and across countries, we eliminate banks with total assets below 1 billion euros at the end of 2014. These filters lead us to our final sample of 83 banks, out of which 30 institutions received public interventions (Table 2).²¹

4.2 Systemic risk variables

The data required for our systemic risk estimations span 2005 to 2014. We chose this period because it allows us to track the evolution of systemic risk during two financial crisis, i.e. the 2008 global financial crisis and the 2010 European sovereign debt crisis. Following Brunnermeier and Oehmke (2013), we estimate systemic risk during the whole period to account for the buildup phase in the precrisis and the propagation phase during the crisis. The impact of policy interventions on systemic risk is analyzed during the same time span. Although there have been important regulatory changes after 2014, most of the bailout packages were provided to European banks during 2008–2012, according to the State Aid Register data of European Commission.

²⁰ For example, several banks from Greece that received government interventions are in this situation and are excluded from the final sample.

²¹ For the assessment of robustness, we re-estimate the empirical specifications for an extended sample of 110 banks, respecting all initial inclusion criteria while also allowing banks of smaller size with total assets between 178 million euros and 1 billion euros. The results are similar to our main findings.

The systemic risk measures are estimated separately for each bank using weekly returns extracted from Datastream (see Table 1 and Online Appendix 4 for computation details). Figure 2 presents the evolution of our sample's market capitalization. This corresponds to the 83 banks analyzed over 521 weeks (2005–2014). Due to the deteriorating economic conditions in international financial markets, market equity shows a downturn in the first phase of the crisis (2008–2009), decreasing by more than 70% in comparison with its maximum value, reached in the middle of 2007. There were signs of recovery during 2009–2010, but the market equity started declining again at the end of 2011 when the European sovereign debt crisis took off.

The summary statistics of the systemic risk indicators are reported in Table 3 (Panel A). The data corresponding to the *Marginal Expected Shortfall* model reveal that for 2005–2014, the quarterly average contribution to systemic risk of all banks translates to an approximately 69% loss of the banks' market capitalization. The statistics resulting from the *Conditional Value at Risk* measure, which we employ as a robustness check, show that banks' marginal contribution to systemic risk represents an approximately 39% loss of the system's market capitalization within a quarter.

The main features are compared between unrescued banks and banks affected by rescue packages (Table 3 Panel B). Overall, the difference in means analysis shows that the mean contribution to systemic risk for the whole sample period is larger for banks that received state guarantees, capital injections or liquidity injections than for nonaffected banks.²²

Figure 3 presents the weekly average contribution to systemic risk of all banks in our sample during 2005–2014. The graph reveals an increase in risk during the Lehman collapse in September 2008 and the European sovereign debt crisis. Our systemic risk measure is very likely to reflect both risk realization and forward-looking risk. Even though the governments of European member states intervened promptly with financial assistance programs, interventions were usually applied after the realization of risk. Therefore, declines in the market equity of bailed-out banks were likely during the period before intervention and may have persisted for a period after. Similarly to us, Black et al. (2016) find that the systemic risk of the European banking system increased during the crisis, reaching a peak during the sovereign debt crisis in Europe.

4.3 Emergency policy interventions

The impact on systemic risk of the emergency measures taken by European member states during the global financial crisis and the European sovereign debt crisis is analyzed for each of several types of policy interventions. To limit the negative spillovers in the banking system and to ensure financial stability, supervisory authorities used a broad range of mechanisms which we group into three categories (1) state guarantees, (2) recapitalizations (capital injections), and (3) liquidity injections. These are described in Online Appendix 2, which provides details on the type, size, and time of implementation of the interventions. Banks from our sample received public guarantees for bond issues, senior notes or other forms of debt; recapitalizations in the form of hybrid capital, participation capital, preferred shares, deeply subordinated perpetual notes or contingent convertible subordinated bonds (CoCos); and liquidity injections consisting of loan facilities, swap facilities, illiquid asset back-up facilities or asset protection schemes (APSs). We hand-collect the dataset from banks' annual reports, financial statements, websites and the State Aid Register of the European Commission.

²² We perform a two-sample t test with unequal variance.

lable 1 Description of variables. Y represents y Worldscope and Datastream, while Own C ^b are c stands for European Central Bank, GFDB for Gl 2011), WDI for World Development Indicators, V and Raddatz (2010)	arry frequency. U is quarterly frequency, and, w is weekly frequency. Own C [*] represents own cat alculations based on data from banks' financial statements, websites and State Aid Register of Eu obal Financial Development Database, SBRS for World Bank Survey of Bank Regulation and Sur WGI for World Governance Indicators, DPI for the Database of Political Institutions of Cruz et al.	utations using data from pean Commission. ECI rrvision (2003, 2007 an- 2016), and BR for Brau
Variable name	Description and calculation	Frequency Source
Dependent variables (bank level)		(
MES	Marginal expected shortfall expressed in units of percentage loss of the bank's market equity within a quarter. MES is defined as in Acharya et al. (2017), i.e., the average return on bank's market capitalization on the weeks the total market capitalization of the sample experi- enced its 1% worst outcomes. The measure is determined using $DCC-GJR$ GARCH method. System is defined by the Market capitalization of the sample. The weekly values are summed up within a quarter. The indicator (loss) is expressed as a positive number, hence higher values denote greater systemic importance	Q Own C ⁴
Delta CoVaR	Contribution to systemic risk expressed in units of percentage loss of the system's market value of equity within a quarter. Delta CoVaR is defined as in Adrian and Brumermeier (2016), i.e., the difference of the Value-at-Risk of the system's market equity conditional on the distress of a particular bank (1% worst outcomes) and the Value-at-Risk of the system's market equity conditional on the median state of the bank. The measure is determined using <i>Quantile Regression</i> , based on weekly market capitalization and a set of domestic and global market indices. The system is the Market capitalization of the sample. The weekly values are summed up within a quarter. The indicator (loss) is expressed as a positive number, higher values denoting greater systemic importance	Q Own C ^a
Data used for estimating systemic risk		
Balance sheet data (bank level)		
Market equity	Market capitalization (bil eur)	W Datastream
Returns on bank i's market equity in week t	$R_{Market Equity (i)} = \frac{Market Equity_{i-1}^{i}}{Market Equity_{i-1}^{i}} - 1 \%$	W Worldscop
Returns on system's market equity in week t	$\begin{split} R^{sys}_{Market \; Equity (t)} &= \sum_i \frac{Market \; Equity_i^{t}}{\sum_i Market \; Equity_i^{t}} \\ \times R^i_{Market \; Equity_i}, i takes values from 1 to the sample's number of banks (\%) \end{split}$	W Worldscop

Table 1 (continued)			
Variable name	Description and calculation	Frequency	Source
Financial market indices			
Government bonds yield	Change in the Euro AAA government bonds yield curve instantaneous forward rate ten-years against one-month residual maturity	м	ECB
Funding liquidity spread	Difference between the Euribor three-month interbank rate and the Euro area government bonds three-month yield curve	M	ECB
Real estate price index	Change in the Real estate price index for Europe	M	Datastream
VSTOXX	Change in the implied volatility index within the Eurozone	M	Bloomberg
Data used for panel regressions			
Policy interventions (bank level)			
State guarantees	Guarantees provided by state j to bank i in quarter t (as % of Total assets). On short run they are maintained at the same level from one quarter before the event to one quarter after the event $(t-1; t+1)$. On long run they are maintained at the same level from one quarter before the event to the quarter they are paid back $(t-1; t+n)$	õ	Own C ^b
Recapitalizations	Capital injections provided by state j to bank i in quarter t (as % of Total assets). On short run they are maintained at the same level from one quarter before the event to one quarter after the event $(t-1; t+1)$. On long run they are maintained at the same level from one quarter before the event to the quarter they are paid back $(t-1; t+n)$	ð	Own C ^b
Liquidity injections	Liquidity injections provided by state j to bank i in quarter t (as % of Total assets). On short run they are maintained at the same level from one quarter before the event to one quarter after the event (t-1; t+1). On long run they are maintained at the same level from one quarter before the event to the quarter they are paid back $(t-1; t+n)$	0	Own C ^b
Regulatory restrictions (bank level)			
Supervisory board intrusions	Dummy variable that takes the value 1 if the government appointed members on the supervisory board and 0 otherwise	0	Own C ^b
Management pay limitations	Dummy variable that takes the value 1 if the government limited the salaries and bonuses of bank's executives and 0 otherwise	0	Own C ^b

$\underline{\textcircled{O}}$ Springer

Table 1 (continued)			
Variable name	Description and calculation	Frequency	Source
Capital payout bans	Dummy variable that takes the value 1 if the government prohibited the dividend and other capital payouts and 0 otherwise	Q	Own C ^b
Bank characteristics (bank level)			
Beta	Covariance of bank i's stock returns with the market's stock returns divided by the variance of the market's stock returns	Ø	Own C ^b
Size	log(Total Assets)	ð	Worldscope
Asset growth	Asset growth relative to average bank asset growth in the bank's country	б	Own C ^a
Leverage	Common Equity/Total Assets (%)	б	Worldscope
Credit risk ratio	Provisions for Loan Losses/Gross Loans (%)	б	Worldscope
Liquidity ratio	Liquid Assets/Deposits and Short-term Funding (%)	б	Worldscope
Rollover risk ratio	Deposits and Short-term Funding/Total Deposits and Borrowings (%)	Q	Worldscope
Return on Average Assets (ROAA)	Net Profit/Average Assets (%)	б	Worldscope
Gross loans share	Gross Loans/Total Assets (%)	б	Worldscope
Net non-interest margin	Net Non-Interest Income/Gross Revenues (%)	б	Worldscope
Dummy size	Dummy variable that takes the value 1 if the size of the bank is above the median size of the intervened banks sample	0	Own C ^a
Dummy leverage	Dummy variable that takes the value 1 if the leverage of the bank is above the median Common Equity to Total Assets ratio of the intervened banks sample	Q	Own C ^a
Dummy profitability	Dummy variable that takes the value 1 if the ROAA ratio of the bank is above the median Net Profit to Average Assets ratio of the intervened banks sample	Q	Own C ^a
Market & Macro controls (country level)			
Competition	Boone indicator, a measure of competition in the banking market calculated as the elasticity of profits to marginal costs. The lower the Boone indicator is, the higher the level of competition	A	GFDB
Capital regulatory index	A composite index that measures the amount of regulatory capital banks must hold and the strin- gency of regulations on the quality capital. The index takes values from 0 (relaxed regulations) to 10 (tight regulations)	A	SBRS
Sovereign debt	Sovereign debt holdings of the banking system as share in GDP	А	ECB

Table 1 (continued)			
Variable name	Description and calculation	Frequency	Source
Inflation	Inflation measured by the consumer price index, reflecting the annual percentage change in the cost to the average consumer of acquiring a basket of goods and services that may be fixed or changed at specified intervals	A	MDI
GDP growth	Gross domestic product at market prices, calculated as $\%$ change on previous period, based on 2005 = 100	Α	MDI
Crisis	Dummy variable that takes the value 1 after the Lehman Brothers collapse and 0 otherwise	0/1	
GFC crisis	Dummy variable that takes the value 1 during 2008 q3 – 2009 q4 and 0 otherwise	0/1	
SVG crisis	Dummy variable that takes the value 1 during 2010 q1 – 2012 q4 and 0 otherwise	0/1	
Exclusion restrictions (country level)			
Political stability index	A composite index that measures the stability of the political institutions. The index takes values from -2.5 (weak) to 2.5 (strong)	A	NGI
Vote share non-government parties	The vote share of parties other than the government ones	A	DPI
Legislative & executive elections	Dummy variable that takes the value 1 if a legislative or executive election took place in the quarter when bailouts were provided	б	DPI
Prevalence	An index that measures the prevalence of connectedness through the ratio of actual to expected number of political connections (i.e., number of cases within a country with a former politician who later sits on board). The probability that a former politician later sits on a bank's board is derived by assuming that the connections are selected randomly with replacement from a com- mon pool. This probability gives the expected number of political connections within a country		BR

172

 $\underline{\textcircled{O}}$ Springer

Country	Number of banks	Total assets sample (billion \in)	Total assets country (billion €)	Total assets sample / Total assets banking system (%)
Austria	6	354.41	915.11	38.73%
Belgium	1	245.17	1021.57	24.00%
Bulgaria	1	4.52	47.41	9.54%
Cyprus	2	34.34	90.20	38.07%
Czech Republic	1	34.39	190.87	18.02%
Denmark	8	576.43	1048.30	54.99%
Finland	1	4.29	525.31	0.82%
France	7	5833.12	7881.63	74.01%
Germany	4	2283.04	7528.95	30.32%
Hungary	1	34.87	116.06	30.05%
Ireland	2	237.26	1016.95	23.33%
Italy	12	1876.55	4047.89	46.36%
Lithuania	1	1.64	24.04	6.82%
Malta	3	16.66	50.33	33.10%
Netherlands	2	1010.12	2250.13	44.89%
Poland	10	221.08	361.63	61.13%
Portugal	3	118.99	515.33	23.09%
Romania	1	8.00	91.40	8.75%
Slovakia	4	24.76	61.13	40.50%
Spain	6	2619.00	3150.74	83.12%
Sweden	4	1476.29	1514.50	97.48%
United Kingdom	3	3047.90	8895.35	34.26%
Total	83	20,062.81	41,344.83	48.53%
EU-28			42,520.53	47.18%

Table 2The distribution of banks. The calculations are based on Worldscope data for Total assets of oursample and European Banking Federation data for Total assets of the banking system in each country atyear end 2014

All three types of bailouts were applied to the banks in our sample during 2008–2014 (Online Appendix 3). In terms of value, liquidity injections lead with an average size of approximately 4% of banks' total assets, followed by state guarantees (3% of total assets) and recapitalizations (2% of total assets). The aim of guarantee schemes is to ensure the supply of liquidity in the interbank market or to prevent bank runs. Recapitalizations are intended to strengthen the capital base of banks. Liquidity injections are given to limit the probability of runs and to encourage bank participation in asset markets, thereby limiting financial instability. They are expected to generate higher liquidity and greater transparency. The guarantees and recapitalizations were more country-wide in nature, as the schemes were aimed at restoring confidence, while the liquidity measures targeted the exposure to losses by individual banks (Panetta et al. 2009).²³

²³ Across our sample, some countries opted for state guarantees (i.e., Cyprus, Sweden) and others for capital injections (i.e., Belgium, France), but countries more frequently opted for both of them. Liquidity injections were more common as standalone interventions.



Fig. 2 Evolution of the sample's market capitalization This figure presents the evolution of the market capitalization for 83 European banks from 2005 to 2014. The values are expressed in billion euros

Table 4 provides descriptive statistics of the financial support provided by the government for the entire sample in Panel A and for the sample restricted to banks with intervention events in Panel B. In sum, our sample was exposed to 106 policy intervention events: 36 events corresponding to state guarantees, 35 events related to recapitalizations, and 35 events linked with liquidity injections (among which 8 are related to APSs and 27 to other types of emergency liquidity schemes). Out of 83 banks, 30 implemented these types of policies: 6 banks received all three types of interventions, 14 banks applied two types of interventions, and 10 banks relied on a single intervention measure. The rescued banks represent 15 countries (out of the 22 included in our sample), among which in seven countries more than two banks were affected. From the 30 rescued banks in our sample, 18 were released from the bailouts before 2014.

In most cases, bailouts come at a cost for banks. For example, banks often paid a fee, usually linked to the risk of the financial institution (i.e., its rating), executive compensation is limited, and dividends could be distributed only to the government (Petrovic and Tutsch 2009). We account for these features by considering a set of restrictions imposed by regulators during the duration of bailouts: supervisory board intrusions, management pay limitations, and capital payout bans. They are expressed as dummy variables, and their definitions are provided in Table 1. Among the policy intervention events from our sample, 28% are associated with supervisory board intrusions, 67% have limitations on executive compensation, and 41% are linked with capital payout restrictions.²⁴

²⁴ From 2013 onward, dividend bans became mandatory for European banks that receive bailouts and are subject to a restructuring plan (EC 2013).

Table 3Summthe whole samwith unequal viaunits of percent	lary stat ple. Pan ariance. tage losi	tistics of nel B pr Definit s of the	f systemic ovides the ion of var system's	risk in differe iables i market	dicator ance in s provid equity v	s. The t means led in ⁷ within t	table re analysi Table 1. a quarte	ports the s betwee . Values 3r (Delta	e summi en non-i are expr (CoVaR	ary stat nterver ressed j	istics of ned bank in units	the dep s and b of percer	endent anks af ntage le	variab fected sss of t	les duri by polic he bank	ng 2005–7 iy interve s' market	2014. Par ntions, be equity w	el A reports the sed on a two-sa ithin a quarter (l	output for mple t-test AES), and,
Panel A. All sampì	le																		
Variables	Unit	4	Mean	Std. de	.v.		Min			p25		p50			p75			Max No	. obs.
MES	Quarterl % loss banks' ket eqi	ly (s of ` Mar- uity	59.30	43.01			0.00			33.75		67.76			100.11			149.09 33	20
Delta CoVaR	Quarterl % loss systen Marke equity	ly is of it is it	38.74	25.26			0.00			19.32		37.69			55.06			92.12 33	20
Panel B. Intervene	d versus 1	non-interv	vened banks																
	Non-inte	ervened b	ank							Interven	ed banks							Difference in mean vened vs. Non-in	s (Inter- tervened)
Variables	Unit	Mean	Std. dev.	Min	p25	p50	p75	Max	No. obs.	Mean	Std. dev.	Min	25 I	50 F	15 1	Max	No. obs.		
State guarantees																			
MES	%	62.32	41.85	0.00	29.00	59.14	89.79	149.09	2560	92.83	38.26	0.00	57.87 9	4.36 1	20.28	149.09	760	30.51	888
Delta CoVaR	%	37.16	25.01	0.00	16.69	37.20	52.91	92.12	2560	44.06	25.40	0.00	24.04 2	2.07	2.31	92.12	760	6.90	000
Recapitalizations																			
MES	%	61.68	41.52	0.00	28.31	58.19	90.83	149.09	2440	90.44	39.91	0.00	55.32 8	8.29	21.38	149.09	880	28.76	888
Delta CoVaR	%	36.96	25.14	0.00	16.22	36.39	53.47	92.12	2440	43.67	24.96	0.00	26.46 4	1.37 5	9.37	92.12	880	6.71	888
Liquidity injection.	s																		
MES	%	63.02	41.60	0.00	29.45	60.23	91.34	149.09	2720	97.78	37.46	0.00	71.70 9	9.62	29.76	149.09	600	34.76	888
Delta CoVaR	%	37.12	24.86	0.00	17.15	36.58	52.99	92.12	2720	46.12	25.76	0.00	25.62 2	4.68	2.37	92.12	600	9.00	8 8 8



Fig.3 Weekly contribution to systemic risk of banks. This figure presents the weekly evolution of systemic risk for all 83 banks in our sample during 2005–2014. Values obtained for each bank are averaged on a weekly base for the whole sample. The output corresponds to the banks' contribution to systemic risk determined via the Marginal Expected Shortfall model. We express the systemic risk indicators as positive numbers (weekly percentage loss of the banks' market capitalization), hence higher values denote greater systemic importance

4.4 Bank-level controls

To account for different risk strategies, we control for the risk profiles of banks one quarter before intervention. Prior studies suggest that bank characteristics such as size, leverage, profitability, and credit and liquidity risk were key drivers of systemic risk during the most recent financial crisis (Demirgüç-Kunt and Huizinga 2010; Tarashev et al. 2010; Acharya et al. 2012; Mayordomo et al. 2014). In line with the literature, the following risk profile indicators are used: (1) size (logarithm of total assets); (2) asset growth (asset growth relative to average bank asset growth in the bank's country); (3) leverage (common equity to total assets ratio and Tier 1 regulatory capital ratio in robustness exercises); (4) the quality of the loan portfolio (provisions for loan losses to gross loans); (5) liquidity ratio (liquid assets to deposits and short-term funding); (6) rollover risk ratio (deposits and short-term funding to total deposits and borrowings); and (7) profitability represented by the ROAA ratio (net profit to average assets). Our presumption is that larger size, asset growth, credit risk, and rollover risk are associated with a higher level of systemic risk, while higher capitalization, liquidity, and profitability are associated with a lower level. Additionally, we capture the orientation of banks' business toward traditional and nontraditional activities by including the share of lending activity (gross loans to total assets) and the net noninterest margin (net noninterest income to gross revenues). Previous studies show that systemic risk is associated

shows statistics for the full sample of 83 bank:	morden a tour t to								
Variables	Unit	Mean	Std. dev.	p25	p50	p75	Min	Max	No. obs.
Policy interventions (bank level)									
A. Full sample									
State guarantees (% of Total assets)	%	0.07	1.67	0.00	0.00	0.00	0.00	7.34	3320
Recapitalizations (% of Total assets)	%	0.03	0.79	0.00	0.00	0.00	0.00	11.67	3320
Liquidity injections (% of Total assets)	%	0.05	1.03	0.00	0.00	0.00	0.00	40.46	3320
Supervisory board intrusions	0/1	0.12	0.33	0.00	0.00	0.00	0.00	1.00	3320
Management pay limitations	0/1	0.24	0.43	0.00	0.00	0.00	0.00	1.00	3320
Capital payout bans	0/1	0.19	0.39	0.00	0.00	0.00	0.00	1.00	3320
B. Intervention events sample									
State guarantees (% of Total assets)	%	2.67	1.77	1.51	1.98	3.44	0.05	7.34	36
Recapitalizations (% of Total assets)	%	1.60	2.10	0.50	1.02	1.95	0.11	11.67	35
Liquidity injections (% of Total assets)	%	4.41	9.13	0.16	0.50	4.30	0.09	40.46	35
Supervisory board intrusions	0/1	0.28	0.45	0.00	0.00	0.00	1.00	1.00	85
Management pay limitations	0/1	0.67	0.47	0.00	0.00	1.00	1.00	1.00	85
Capital payout bans	0/1	0.41	0.50	0.00	0.00	0.00	1.00	1.00	85
Risk profile indicators (bank level)									
Beta	Ι	-0.04	0.35	-0.16	-0.03	0.07	-4.99	2.80	3283
Size	Log(bil. €)	3.74	2.16	2.14	3.62	5.50	-0.46	7.60	2381
Asset growth	%	0.43	3.80	-0.21	0.21	0.85	-18.03	21.95	2371
Leverage	%	7.40	3.27	4.91	6.88	9.13	1.08	24.50	2381
Credit risk ratio	%	0.89	1.02	0.32	0.65	1.17	-1.57	13.87	2361
Liquidity ratio	%	32.43	28.97	13.61	23.65	40.44	2.31	223.38	2379
Rollover risk ratio	%	77.17	17.53	63.87	78.83	93.48	14.15	100.00	2366
ROAA	%	0.62	0.95	0.24	0.55	0.93	-10.46	10.64	2394
Gross loans share	%	61.12	16.20	53.95	63.70	72.99	3.34	90.39	2378
Net non-interest margin	%	-22.91	35.11	-31.29	-20.47	-10.22	-859.52	808.46	2360

 $\underline{\textcircled{O}}$ Springer

Table 4 (continued)									
Variables	Unit	Mean	Std. dev.	p25	p50	p75	Min	Max	No. obs.
Market & Macro conditions (country level)									
Competition (Boone indicator)	I	-0.05	0.06	-0.09	-0.04	-0.03	-0.20	0.22	2988
Capital regulatory index	I	5.98	2.03	4.00	6.00	8.00	3.00	11.00	3320
Sovereign debt	I	68.00	29.59	44.27	64.52	86.06	11.95	135.37	3320
Inflation	%	2.11	1.47	1.22	2.06	2.81	-4.48	12.35	3320
GDP growth	%	0.94	3.06	-0.63	1.30	2.55	-13.86	12.41	3304
Exclusion restrictions (country level)									
Political stability index	I	0.74	0.39	0.50	0.84	1.04	-0.47	1.59	3320
Vote share of non-government parties	%	0.94	3.00	0.00	0.00	0.00	0.00	23.40	3240
Legislative & executive elections	0/1	0.09	0.30	0.00	0.00	0.00	0.00	2.00	3320
Polarization	I	1.25	0.91	0.00	2.00	2.00	0.00	2.00	3008

178

with a high share of nontraditional activity (Brunnermeier et al. 2020; Demirgüç-Kunt and Huizinga 2010). Additionally, we capture the comovement of bank value with financial system value by including the beta coefficient, computed as the covariance of bank i's stock returns with the market's stock returns divided by the variance of the market's stock returns. The variables are extracted from Worldscope, and their definitions are given in Table 1.²⁵ The descriptive statistics presented in Table 4 show that on average during 2005–2014, banks from our sample had a Tier 1 ratio of 11%, liquidity ratio of 32%, credit risk ratio of 1% and gross loans shares of 61%. These statistics suggest that on average, the institutions are well capitalized and have a good liquidity situation and a high-quality loan portfolio. Additionally, they are more oriented toward traditional banking business.

4.5 Macro controls

Following previous studies (Girardi and Ergün, 2013; Anginer et al. 2014a; Weiß et al. 2014a, b), we control for the banking market and macroeconomic environment. Accounting for the particularities of each national banking sector, we consider the intensity of competition – or lack thereof – expressed by the Boone indicator,²⁶ which has been found to be associated with a reduced contribution to systemic risk (Anginer et al. 2014a). Next, we account for the strictness of prudential regulations regarding initial and overall capital held by banks. As a proxy, we use the capital regulatory index provided by the Bank Regulation and Supervision Database of the World Bank and calculated as in Barth et al. (2013).²⁷ Additionally, we employ the sovereign debt holdings of the banking system as a share in GDP and the inflation rate and GDP growth as macro controls. We expect systemic risk to increase with the deterioration of macro conditions. Finally, we include in our specifications a dummy variable that reflects the global financial crisis (Dummy GFC) and a dummy variable that accounts for the European sovereign debt crisis (Dummy SVG) to control for the dynamics of these crises that affected the evolution of systemic risk. The variables are extracted from the World Development Indicators, Global Financial Development and European Central Bank databases. Their definitions are given in Table 1, and the descriptive statistics are given in Table 4.

5 Empirical results

This section presents the results of the regressions with the systemic risk measures as our dependent variables and policy interventions as the main determinants. First, we discuss the influence of policy interventions on banks' systemic importance. Both short- and long-term effects are assessed. Second, we examine how regulatory restrictions and banks' risk strategies affect the impact of emergency rescue actions on systemic risk. The main regressors of interest include the volume of emergency rescue packages received by bank i from

²⁵ In the regression analysis, several of these variables are transformed to assure stationary behavior, as indicated by panel unit root tests. Table 1 gives the transformation formulae.

²⁶ The Boone indicator is a measure of competition in the banking market, calculated as the elasticity of profits to marginal costs. The lower the Boone indicator is, the higher the level of competition.

 $^{^{27}}$ Because these data are available only for years 2007 and 2012, we maintain their level constant for the period 2008–2011 (the 2007 level), and for the period 2013–2014 (the 2012 level). The capital regulatory index takes values from 0 (relaxed regulations) to 10 (tight regulations).

government j during the event window as a share of banks' total assets. In the case of shortrun estimations, these bailouts take the same value from one quarter before intervention to one quarter after. In the case of long-run estimations, the bailouts take the same value from one quarter before intervention to the quarter in which they are paid back.

5.1 Emergency policy interventions and systemic risk: Baseline results

The output for the 1st-stage Heckman (1979) selection model is presented in Table 5 Panel A. The results reveal that greater political stability significantly increases the probability of the government providing bailouts to financial institutions. The vote share of nongovernment parties enters the model significantly, with a positive sign, suggesting that in countries with higher political competition where the vote share of parties other than the government parties is higher, the regulator is more likely to implement policy interventions. Prevalence is significant as well. The F statistic associated with the pseudo Kleibergen–Paap F test indicates that the set of identifying covariates has sufficient predictive power to explain the likelihood of the government providing financial assistance.²⁸ Additionally, the p value corresponding to the pseudo Hansen J test validates the orthogonality between the exclusion restrictions and systemic risk. Therefore, our set of instruments contains additional information on the likelihood of bailouts and meets the validity conditions.²⁹

5.1.1 Short-term effects

Table 5 Panel B shows the estimation results for the *OLS fixed effects* regression presented in Eq. (2). The dependent variable captures the banks' contribution to systemic risk, estimated using the *Marginal Expected Shortfall* methodology. Model (2) provides a baseline specification that includes all policy interventions, bank characteristics, micro and macro controls, bank fixed effects and year fixed effects.³⁰ Model (3) includes additional country × year fixed effects, and Model (4) includes bank fixed effects and country × year fixed effects.

We consider Column (4), which includes all bank and macro characteristics, as well as bank fixed effects and country \times year fixed effects, our benchmark specification. In what follows, the interpretation of the empirical results is detailed for this model.³¹ A

²⁸ The rule of thumb is to use the Stock and Yogo (2005) critical values. Specific to our sample, the Stock– Yogo weak identification test critical value for 5% maximal relative bias is 16.85.

²⁹ Because our panel specifications (2^{nd} stage) include all three policy interventions together, we have to use a single estimate of the inverse Mills Ratio from the 1^{st} -stage Probit equation. Therefore, we estimate the impact of the exclusion restrictions on the probability of a bank receiving a bailout, regardless of the bailout type. In robustness exercises, we rerun Eq. (1) using as a dependent variable the probability of a bank receiving each type of bailout separately. The (unreported) results show that the relation between the exclusion restrictions and each type of bailout is similar with that in the main findings in terms of sign, size, and significance.

 $^{^{30}}$ We also include the policy intervention variables separately in the empirical specifications. The (unreported) results show that the sign, size, and significance of the coefficients remain unchanged. Out of 30 rescued banks in our sample, six of them received all three types of interventions, while 14 banks were subject to two types of interventions and 10 relied on a single type of bailout. Because the correlation coefficients among the three types of bailouts are low (between 0.004 and 0.022), we report the results for the specifications that consider all types of bailouts together.

³¹ A limitation of our analysis is that causality cannot be established with certainty due to the lack of a natural experimental setting within our framework. To address this issue, we provide alternative explanations to the maximum extent possible.

IADIC 2 Short-ferm and long- the first stage results from a <i>P</i> (<i>1</i> — <i>P</i>) <i>mervened</i> ii, <i>j</i>)) = $b_0+\Omega \times Ia$ tifying restrictions (political st from 22 European countries at via <i>OLS Fixed Effects: System</i> The sample is restricted to 30 in column (1). The dependent policy interventions received 1 effects (i.e., interventions take same value within the event w Explanatory variables are one Table 1. Different specificatior	term impact or p robit model used <i>inifying restricti</i> tability, vote shar and the period acc in <i>Risk</i> _{ijt} = $\beta_0 + \beta_1$ i) banks that recei variable is represe by bank i (i.e., in z the same value f vindow (t-1; t+1) reperiod lagged. A ns include year fix	oucy intervent to estimate th ions _{iji-1} + $\Phi \times$ e of non-gove ounts for 40 q ounts for 40 q ounts for 40 q i × Policy inte ved interventi sented by ME terventions tr from one quan) and the long Il models incl	tutions by bank e probability (Bank controls, trannent partie: quarters during quarters during quarters during ions and contro S and reflects ions and reflects ake the same v reter before the strun effects of lude an unrepo ink fixed effect	s on systemic of bank i to red $g_{1L,1} + \Psi \times Marn$ s, legislative a s, 2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F (2005–2014. F) (2005–2014. F) (2	risk. Inis tab risk. Inis tab reseive policy in ket & Macro and executive of anels B, C, an and $MR_{ij,i-1} + \boldsymbol{\phi} \times$ ple selection × ple selection × ple selection × ple selection × ple selection × platter they u entions (i.e., i variables are × year fixed ef	the reports the just reports the interventions in controls $j_{ijk+1} + \iota$, elections dumm and D report the Bank controls, bias by includis bias by includis i from country interventions ta s winsorized at fects. Standard	results of the <i>I</i> quarter <i>t</i> , using quarter <i>t</i> , using quarter <i>t</i> , using $\gamma_i + \varepsilon_{ijr}$. The end $\gamma_i + \varepsilon_{ijr}$. The end γ_i , ρ_i ,	Teckman Selec g the following ppirical specific titions of the fol titions of the fol et & Macro cc Mills ratio ge Panel B consi fiter the event), h the short-rur alue within the ^h percentiles. T	thom Model. Paramon Model. Paramon Model. Paramon Largession: Lication includes: the full samp llowing regress $puttors_{j_i, i_j} + q_i$, $puttors_{j_i, i_j} + q_i$, $puttors_{j_i, i_j} + q_i$, while Panel (while Panel (i.e., interver event window Their definition evel are report	In the Horizon equation of the second secon
	A. Identi- fication 1 st stage	B. Short rui 2 nd stage	n (t-1; t + 1)		C. Long ru 2 nd stage	ın (t-1; t + n)		D. Short run (t+2; t+n) 2^{nd} stage	ı (t-1; t+1) and	l long run
	Probit	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS
Dependent variable	Intervened (1)	MES (2)	MES (3)	MES (4)	MES (5)	MES (6)	MES (7)	MES (8)	MES (9)	MES (10)
Policy interventions										
Guarantees short run		6.546^{***} (1.838)	7.646*** (2.137)	7.534*** (2.211)				6.481^{***} (1.834)	7.289*** (2.180)	7.243*** (2.395)
Recapitalizations short run	_	4.864	3.140	2.825				2.439	1.634	0.491
		(3.574)	(3.323)	(3.216)				(3.213)	(2.945)	(2.920)
Liquidity injections short ru	un.	0.218	0.190	0.107				0.316	0.236	0.337
		(0.379)	(0.494)	(0.559)				(0.423)	(0.484)	(0.489)
Guarantees long run					1.148 (1.213)	1.282 (1.184)	0.567 (1.508)	0.781 (1.086)	0.849 (1.199)	0.115 (1.190)
Recapitalizations long run					-2.493	-3.049**	-4.248* 17.148)	-4.710**	-5.226**	-7.522** 17.076)
					(000.7)	(1001)	(0+1.7)	(+-(1)-2)	(00((016.7)

Table 5 (continued)										
	A. Identi- fication 1 st stage	B. Short rur 2 nd stage	1 (t-1; t + 1)		C. Long run 2 nd stage	(t-1; t+n)		D. Short run (t+2; t+n) 2^{nd} stage	(t-1; t+1) and	l long run
	Probit	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS
Dependent variable	Intervened (1)	MES (2)	MES (3)	MES (4)	MES (5)	MES (6)	MES (7)	MES (8)	MES (9)	MES (10)
Liquidity injections long run					0.876*** (0.278)	1.071*** (0.228)	1.084^{***} (0.182)	0.768** (0.320)	1.026*** (0.232)	0.979*** (0.212)
Identification										
Political stability	1.366*** (0.126)									
Vote share of non-govern- ment parties	0.070^{***} (0.011)									
Legislative & executive elections	0.050 (0.111)									
Prevalence	-0.517^{***} (0.071)									
Inverse Mills ratio		1.819 (14.408)	100.390*** (31.401)	102.529*** (36.462)	2.573 (13.015)	87.194*** (30.884)	98.141*** (34.015)	-0.759 (12.995)	87.061*** (31.803)	91.915** (35.878)
Bank characteristics										
Beta	0.328^{*} (0.188)	-10.787 (7.041)	2.591 (7.704)	5.422 (8.848)	-10.540* (5.958)	-2.329 (7.653)	3.233 (8.029)	-10.122 (6.179)	-2.068 (7.631)	2.850 (8.387)
Size	0.503^{**} (0.033)	8.926 (9.533)	38.093*** (8.126)	38.157** (15.445)	7.512 (8.226)	34.366*** (8.042)	41.906*** (10.632)	9.925 (8.419)	34.014*** (8.300)	44.072*** (11.867)
Asset growth	0.000 (0.011)	0.062 (0.327)	0.239 (0.432)	0.227 (0.421)	0.076 (0.364)	0.178 (0.451)	0.214 (0.439)	0.038 (0.305)	0.076 (0.422)	0.121 (0.408)
Leverage	-0.032* (0.017)	-3.503** (1.619)	-3.169*** (1.199)	-4.095* (2.075)	-4.681** (1.703)	-3.154** (1.308)	-4.570* (2.270)	-3.395** (1.591)	-3.275** (1.386)	-3.726 (2.193)

Table 5 (continued)										
	A. Identi- fication 1 st stage	B. Short run 2 nd stage	1 (t-1; t+1)		C. Long run 2 nd stage	(t-1; t + n)		D. Short run (t + 2; t + n) 2^{nd} stage	(t-1; t+1) and	long run
	Probit	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS
Dependent variable	Intervened (1)	MES (2)	MES (3)	MES (4)	MES (5)	MES (6)	MES (7)	MES (8)	MES (9)	MES (10)
Credit risk ratio	0.214***	4.879	11.235*	11.376	4.836	9.987*	10.729*	2.973	8.738	8.837
	(0.056)	(4.178)	(6.204)	(6.667)	(4.186)	(6.059)	(6.202)	(4.074)	(6.129)	(6.528)
Liquidity ratio	-0.005** (0.002)	-0.144 (0.123)	-0.352*** (0.126)	-0.348^{***} (0.119)	-0.092 (0.120)	-0.333** (0.131)	-0.321*** (0.108)	-0.068 (0.124)	-0.318** (0.126)	-0.311^{**} (0.115)
Rollover risk ratio	-0.016^{***}	-0.078	-0.681	-0.965^{*}	0.081	-0.542	-0.911*	0.057	-0.487	-0.864*
	(0.004)	(0.231)	(0.443)	(0.479)	(0.208)	(0.431)	(0.446)	(0.208)	(0.444)	(0.470)
ROAA	0.115	0.380	3.095	3.148	1.134	3.250	3.495	-1.067	2.424	1.860
	(0.078)	(3.267)	(2.852)	(3.149)	(3.345)	(2.873)	(3.219)	(3.002)	(2.795)	(3.344)
Gross loans share	0.022***	-0.624*	0.643	1.050	-0.511	0.408	1.076	-0.748**	0.404	0.809
	(0.004)	(0.324)	(0.399)	(0.659)	(0.331)	(0.400)	(0.686)	(0.332)	(0.423)	(0.641)
Net non-interest margin	-0.007^{***}	-0.226**	-0.416^{***}	-0.387^{***}	-0.261^{***}	-0.400^{***}	-0.395 ***	-0.237**	-0.385^{***}	-0.365**
	(0.003)	(0.092)	(0.125)	(0.133)	(0.086)	(0.123)	(0.128)	(0.091)	(0.129)	(0.138)
Market and macro controls										
Competition	0.012^{***}	-0.156	2.365***	-1.550^{**}	-0.186	2.229***	-0.948	-0.191	2.168***	-1.638**
	(0.003)	(0.279)	(0.806)	(0.704)	(0.212)	(0.767)	(0.853)	(0.261)	(0.798)	(0.772)
Capital regulatory index	0.033*	-1.973^{**}	-6.230***	-13.742***	-2.359***	-5.736**	-14.587***	-2.077***	-6.116^{**}	-14.487***
	(0.019)	(0.836)	(2.279)	(2.137)	(0.582)	(2.278)	(4.225)	(0.692)	(2.341)	(2.122)
Sovereign debt	0.008***	0.859***	-0.226	-0.532^{*}	0.828***	-0.228	-0.448	0.937^{***}	-0.168	-0.306
	(0.002)	(0.172)	(0.495)	(0.262)	(0.159)	(0.485)	(0.342)	(0.163)	(0.499)	(0.237)
Inflation	0.169^{***}	1.198	3.116	-7.116*	2.824	9.163***	-2.866	2.599	8.706***	-8.972**
	(0.050)	(2.896)	(3.191)	(3.848)	(2.736)	(2.816)	(4.464)	(2.569)	(3.063)	(4.280)
GDP growth	-0.117^{***} (0.026)	-0.037 (2.025)	-9.344^{***} (2.181)	8.532*** (2.174)	1.330 (1.898)	-6.599*** (2.220)	8.135*** (2.202)	1.401 (1.964)	-6.950*** (2.099)	8.792*** (2.055)

Table 5 (continued)										
	A. Identi- fication 1 st stage	B. Short run 2 nd stage	(t-1; t+1)		C. Long run 2 nd stage	(t-1; t+n)		D. Short run (t+2; t+n) 2^{nd} stage	(t-1; t+1) an	d long run
	Probit	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS	OLS
Dependent variable	Intervened (1)	MES (2)	MES (3)	MES (4)	MES (5)	MES (6)	MES (7)	MES (8)	MES (9)	MES (10)
GFC crisis	-0.167 (0.250)	18.492*** (5.277)	13.669** (6.621)	13.025* (7.216)	20.077*** (5.471)	16.019** (6.573)	14.083* (7.364)	18.932*** (5.305)	15.325** (6.398)	13.958* (7.057)
SVG crisis	-0.864^{***} (0.179)	21.954** (8.761)	-34.044 (21.008)	-0.008 (5.223)	25.984*** (7.591)	– 19.990 (19.918)	9.762* (5.498)	22.774** (8.694)	-15.794 (21.172)	-8.099 (7.380)
Year FE	YES	YES	NO	NO	YES	NO	NO	YES	NO	NO
Bank FE	NO	YES	NO	YES	YES	NO	YES	YES	NO	YES
Country × Year FE	ON	NO	YES	YES	NO	YES	YES	NO	YES	YES
Cluster	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks
Observations	1,831	710	710	710	710	710	710	710	710	710
Number of banks	83	30	30	30	30	30	30	30	30	30
No of countries	22	22	22	22	22	22	22	22	22	22
Within R-squared		0.469	0.575	0.578	0.459	0.566	0.570	0.480	0.584	0.588
Pseudo R-squared	0.343									
Log-likelihood	-809.9									
Kleibergen-Paap F-test statistic	36.316									
Hansen J test statistic	1.556									
Hansen J p-value	0.6695									
*, ** and *** denote significance	the second se	, 5% and 1%								

184

negative coefficient is related to a lower systemic importance, while a positive coefficient is linked to an increased contribution of banks to systemic risk.³²

Overall, the results specific to the short run validate the "delayed effects" hypothesis and indicate that interventions do not succeed immediately. There are several possible explanations for our findings. First, as governments usually provide bailouts after risk is realized, declines in the market equity of bailed-out banks are likely before intervention and may persist for a period thereafter. This leads to a higher measured systemic risk contribution, considering the nature of our MES variable, which captures risk realizations as well as forward-looking risk. Second, such delayed success might have arisen due to some European states' lack of credibility in providing financial assistance during the GFC and the ESDC. Moreover, the imposition of regulatory constraints, behavioral commitments, or fees during the duration of interventions would tend to make the coefficients more positive.

The findings show strong evidence that the provision of *state guarantees* is associated with enhanced systemic risk in the short run. A one-standard-deviation increase in the guarantees offered by the state normalized by total assets is linked with an increase in the systemic contribution of banks by approximately 27% of its standard deviation (as measured by *MES*). Given that the mean contribution to systemic risk is approximately 69% (i.e., the quarterly percent loss of the banks' market capitalization during 2005–2014), the corresponding semielasticity is 17%. This finding is in line with our hypothesis that the credibility of governments' debt guarantees was affected during the period analyzed, which could be associated with declines in bank equity values. These results can be linked with other evidence in the literature showing that guarantees from public funds can severely distort financial stability. This might happen due to the constraints on bailout budgets imposed by tight fiscal requirements or due to the lack of credibility related to their efficiency (Honohan and Klingebiel 2003; Laeven and Valencia 2012; Allen et al. 2015).

Recapitalizations also relate positively to banks' systemic importance, but the effect is not significant. A possible explanation is that even though recapitalizations improve bank solvency, they also dilute the preintervention shareholders' earnings and voting rights in most cases, which is likely to make the coefficients more positive. Additionally, the restrictions associated with recapitalizations are more severe than those involved in other bailouts (i.e., dividend bans, government interference in bank management), and the regulatory fees are higher, which can reduce the market valuation of the rescued banks.³³

Liquidity injections are linked positively to systemic risk, but the effect is not notable in the short run. Usually, the liquidity lines have a window during which banks can benefit from the support; thus, they take longer to produce any effects. Additionally, they were given at lower stages of distress (i.e., a higher level of capitalization) than other types of financial aid (Berger et al. 2022) and in later rounds, as supplementary assistance after recapitalizations or state guarantees were already in place (Brei and Gadanecz 2012), which might explain why they do not carry so much weight in the short run.

The inverse Mills ratio is highly significant, which indicates that selection is present with respect to interventions. Among bank characteristics, the findings suggest that banks' size, leverage, return on assets, and net noninterest margin are key drivers of their systemic risk contribution, in line with the findings from previous literature (Adrian and Brunnermeier 2016; Girardi and Ergün 2013; Anginer et al. 2014a). For banking market characteristics, the capital regulatory index enters the specifications with a negative sign, suggesting that tight prudential regulations regarding initial and overall capital held by banks help decrease their systemic importance.

³² The dependent variable is expressed in units of percentage loss of the banks' market capitalization within a quarter.

³³ Additional results on the effects of restrictions are presented in subsection 5.2.

5.1.2 Long-term effects

In Table 5 Panel C, we analyze the long-run impact of policy interventions, considering that bailouts take the same value from one quarter before the event to the quarter in which they are paid back. The repayment period varies across banks, from one year to six years. The empirical results presented in our benchmark model (Table 5 Column (7)) reflect a positive effect of *state guarantees* in the long run, but the associated coefficient is not statistically significant. This might be possible because debt guarantee support was sometimes difficult to trust. Many European governments encountered a deterioration in fiscal capacity throughout the period analyzed in Europe, which affected the credibility of their guarantees of debt issued by banks.

In the longer-term periods, *recapitalizations* are associated with reduced systemic importance, but the effect is only marginally significant and present only in Models (6) and (7). A one-standard-deviation increase in equity injected by government normalized by total assets is associated with a decrease in banks' contribution to systemic risk by 12% of its standard deviation. The corresponding semielasticity is approximately 8%. Considering that the long-term period in our framework also partly captures the realization of risk, this finding suggests that recapitalizations fix the problems created by risk realizations.

Liquidity injections provided by governments end up being positively and significantly associated with systemic risk, as suggested by the positive and significant coefficients. The economic effect is also meaningful. A one-standard-deviation increase in liquidity injections reported to total assets corresponds to an increase in banks' marginal contribution to systemic risk by approximately 19% of its standard deviation (as measured by *MES*). Our estimates imply an associated semielasticity of 12%. This result indicates that the realization of distress continued at banks for which liquidity injections were in place in the long run. The financial health of these banks remained weak, and their stock returns underperformed.

In Panel D, we include both the short-run and the long-run effects in the same regression. For the short-run effects, we use the same definition of the event window (i.e., interventions take the same value within the window (t-1; t+1)). However, for the long-term effects of policy interventions, we exclude the short-term period from the event window (i.e., interventions take the same value within the window (t+2; t+n)). As shown by Columns (8)–(10), the coefficients are similar to those in the main findings.

In Table 6, we explore additional long-term strategies. We examine the effects of interventions after different time intervals, as it might be expected that the implications of bailouts vary in the long run. To account for these possible developments, we consider several event windows: one year (i.e., policy interventions are maintained at the same level from one quarter before the event to four quarters after the event) in Column (1), two years in Column (2), three years in Column (3), and four years in Column (4). The specifications are similar to the benchmark model from Table 5 Column (7), which includes bank-level, market, and macro controls, bank fixed effects and country×year fixed effects. In Column (5), we report the results of an empirical specification that includes in the same regression individual effects for each year, i.e., the 1st-year effects (t-1; t+4), the 2nd-year effects (t+5; t+8), the 3rd-year effects (t+9; t+12), and the 4th-year effects (t+13; t+16). The results point to the same positive association of liquidity injections with banks' contribution to systemic risk. State guarantees are not significantly associated with systemic importance over the period analyzed, while recapitalizations are significantly linked with reduced systemic importance only for the second year.

Table 6 Additional long-term effects. This table rel $R_{ij,i-1} + \Phi \times Bank$ controls $y_{ij-i} + \Psi \times Market \& Macros Fixed Effects. The sample is restricted to 30 banks t stage Probit model. The dependent variable is representing programment j in quarter take the same value the same regression individual effects for each year, (t + 13; t + 16). The coefficients for bank characterimargin), and market and macro control variables (brevity. Explanatory variables are the 1^{st} and 99th percentiles. Their definitions are gate$	sports the estimation result or controls, $j_{i,j} + \varphi_i + \mu_{ji} + e_i$, that received intervention, esented by MES and reflec from one quarter before t i, i.e. the 1 st -year effects (I isitics (beta, size, asset grr (competition, capital regu ed. All models include an given in Table 1. Standard	Is of the following regressing the results $f_{j,r}$. We report the results f is and control for the same scatter he systemic risk of the intervention to one-for the intervention to one-for -1 ; $t + 4$), the 2^{n_d} year effit risk owth, leverage, credit risk inlatory index, sovereign of unreported constant, couther reported at the bold errors clustered at the bold of the source of the bold of the bold of the source of the bold of the source of the bold of	ssion: <i>Systemic Risk</i> _{<i>ij</i>,1} = β_0 , or the benchmark model (<i>i</i>) the selection bias by inclu ank i from country j in qu ark i from country j in qu ark i from country j in qu ark i from country j in qu they are after the intervent country, rollover risk, lebt, inflation, GDP grow intry × year fixed effects a ank level are reported in bi	$+\beta_1 \times Policy interventions_{i.e.}, Column (7) Table 5).i.e., Column (7) Table 5).ding the inverse Mills raticarter t. Policy interventioranter t. Policy interventiorion in columns (1)-(4). Ccion in col$	terror window + $\beta_2 \times IM$ Method used is OLS generated by the 1 st s received by bank i lumn (5) includes in thm 4 th -year effects are, net non-interest) are suppressed for tables are winsorized
	1 year effects (t-1; t+4)	2 years effects (t-1; t+8)	3 years effects (t-1; t+12)	4 years effects (t-1; t+16)	1–4 years effects indi- vidually
	OLS 2 nd stage	OLS 2 nd stage	OLS 2 nd stage	OLS 2 nd stage	OLS 2nd stage
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)	MES (5)
Policy interventions					
Guarantees long run	-0.391 (1.268)	-2.371 (1.761)	1.129 (1.527)	0.018 (1.401)	
Recapitalizations long run	0.234	-9.679***	-0.175	-2.525	
	(2.858)	(2.191)	(3.150)	(2.388)	
Liquidity injections long run	0.420 (0.307)	0.985^{***} (0.185)	0.872^{***} (0.233)	1.064^{***} (0.180)	
Guarantees long run 1^{st} year (t-1; t + 4)					-3.830 (2.383)
Recapitalizations 1 st year (t-1; t+4)					-6.141
					(3.929)
Liquidity injections 1 st year (t-1; t+4)					0.870^{***} (0.250)
Guarantees long run 2^{nd} (t + 5; t + 8)					-1.290
					(2.163)
Recapitalizations 2^{nd} year $(t + 5, t + 8)$					-15.202*** (3.693)
					(2512)

Table 6 (continued)					
	1 year effects (t-1; t+4)	2 years effects (t-1; t+8)	3 years effects (t-1; t+12)	4 years effects (t-1; t+16)	1–4 years effects indi- vidually
	OLS 2nd stage	OLS 2 nd stage	OLS 2 nd stage	OLS 2 nd stage	OLS 2 nd stage
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)	MES (5)
Liquidity injections 2 nd year (t+5; t+8)					1.543*** 0.467)
Guarantees 3^{rd} year $(t+9; t+12)$					2.032
•					(2.293)
Recapitalizations 3^{rd} year (t + 9; t + 12)					3.177
Liquidity injections 3^{rd} year $(t + 9; t + 12)$					1.884**
((0.780)
Guarantees 4 th year (t+13; t+16)					0.407 (1 490)
Recapitalizations 4 th year (t+13; t+16)					-2.996
					(2.759)
Liquidity injections 4 th year (t+13; t+16)					1.367 * * (0.327)
Identification					
Inverse Mills ratio	102.356*** (35.700)	84.413** (33.930)	103.278^{***} (34.755)	98.882^{***} (34.363)	89.048 ** (34.613)
Bank characteristics	YES	YES	YES	YES	YES
Market and macro controls	YES	YES	YES	YES	YES
Bank FE	YES	YES	YES	YES	YES
Country × Year FE	YES	YES	YES	YES	YES
Cluster	Banks	Banks	Banks	Banks	Banks
Observations	710	710	710	710	710

🙆 Springer

(continued)
Table 6

	1 year effects	2 years effects	3 years effects	4 years effects	1-4 years
	(t-1; t+4)	(t-1; t+8)	(t-1; t+12)	(t-1; t + 16)	effects indi- vidually
	OLS 2 nd stage	OLS 2nd stage			
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)	MES (5)
Number of banks	30	30	30	30	30
No of countries	22	22	22	22	22
Within R-squared	0.562	0.579	0.566	0.568	0.601

 $^{*}, ^{**}$ and *** denote significance levels of 10%, 5% and 1%

5.1.3 Robustness

We check the robustness of our results by employing several strategies related to the sample, period, and methodology used.³⁴

Sample Some of the banks in our sample were severely hit by the European sovereign debt crisis, during which different rescue packages were provided. Additionally, the Eastern and Western EU member states were affected differently in terms of fiscal capacity. To assess the effects of the bailouts across different EU member states, we employ two empirical strategies. Table 7 Panel A presents the output for a restricted sample, where banks from countries severely affected by the European sovereign debt crisis (Cyprus, Ireland, Italy, Portugal, and Spain) are excluded. The findings linked to the short-run impact remain unaltered. For the long-run impact, liquidity injections maintain their positive and significant association with systemic risk. In Table 7 Panel B, we re-estimate the benchmark models for a sample restricted to banks with headquarters in Western Europe.³⁵ As shown by Columns (3) and (4), the size and significance of the coefficients remain very similar to those in the main findings.

Period The main results reported in Table 5 show that systemic risk was significantly higher during the crisis period. To disentangle the effects of interventions on banks during the two crisis episodes covered by our dataset, we interact the policy interventions with dummy variables that reflect the duration of the crises.³⁶ The crisis period started in 2008Q3 and ended in 2012Q4 and consists of two phases. The first phase of the crisis began after the Lehman Brothers collapse in 2008Q3 and continued through 2009Q4, corresponding with the intensification of global financial crisis effects in Europe (Brei et al. 2013). The second phase of the crisis ran from 2010Q1 to 2012Q4 and coincided with the European sovereign debt crisis of Greece, Ireland, Italy, Portugal, and Spain (De Santis 2014).

The results from Table 8 point to the same positive and significant association of state guarantees with systemic risk in the short run and of liquidity injections in the long run but no significant effect of recapitalizations. The long-run impact of state guarantees on systemic risk is negative. For the global financial crisis period, it becomes even more negative and significant (Model 2), suggesting that guaranteeing the debt issued by banks was an efficient intervention strategy during this period. In turn, for the sovereign debt crisis period, the long-run beneficial effect of state guarantees on systemic risk is diminished

```
\beta_3 \times Period_i + \beta_4 \times IMR_{ij,i-1} + \Phi \times Bank \ controls_{ij,i-1} + \Psi \times Market \ \& \ Macro \ controls_{j,i-1} + \varphi_i + v_{ji} + \varepsilon_{ij,i-1} + \varphi_i + v_{ji} + v_{ji}
```

 $^{^{34}}$ For brevity, we present the output for the short-run benchmark model (Table 5 Column (4)), and for the long-run benchmark specification (Table 5 Column (7)). Unreported results show that the findings are similar across the other empirical specifications.

³⁵ Due to the reduced number of interventions provided to Eastern European banks, we cannot run this empirical exercise for a sample restricted to these countries, and hence, we cannot entirely exclude the possibility that the patterns that we observe can only be found in Western Europe.

³⁶ We estimate the following regression: SystemicRisk_{ij,t} = $\beta_0 + \beta_1 \times Policy$ interventions_{ij, event window} + $\beta_2 \times Policy$ interventions_{ij, event window} $\times Period_t + \beta_2 \otimes Policy$

where $Period_t$ is a dummy variable that takes alternatively the value one during the global financial crisis or, respectively, during the European sovereign debt crisis. The description of the other regressors is similar to that for Eq. (2).

Table 7 Robustness assessment: sample. Panel A presents the results for a sample of banks from countries that were not affected by the European sovereign debt crisis. Panel B depicts the output for a sample restricted to Western European banks. Models (1) and (3) assess the short-term impact of policy interventions received by bank i from government j (i.e., interventions take the same value from one quarter before the event to one quarter after the event). Models (2) and (4) assess the long-term impact of policy interventions (i.e., interventions take the same value from one quarter before the event to the quarter they unwind). We report the results for the benchmark models (i.e., Column (4) Table 5 for short run effects, and, respectively, Column (7) Table 5 for long run effects. Method used is OLS Fixed Effects. The estimations control for the sample selection bias by including the inverse Mills ratio generated by the 1st stage *Probit model*. The dependent variable is the MES. The coefficients for bank characteristics (beta, size, asset growth, leverage, credit risk, liquidity, rollover risk, profitability, gross loans share, net non-interest margin), and market and macro control variables (competition, capital regulatory index, sovereign debt, inflation, GDP growth, GFC crisis, SVG crisis) are suppressed for brevity. Explanatory variables are one period lagged. All models include an unreported constant, country × year fixed effects and bank fixed effects. Variables are winsorized at the 1st and 99th percentiles. Their definitions are given in Table 1. Standard errors (S.E.) clustered at bank level are reported in brackets

	A. Without ba affected by the debt crisis	nks from countries e European sovereign	B. Without bar European cour	nks from Eastern htries
	Short run $(t-1; t+1)$	Long run $(t-1; t+n)$	Short run (t-1; t+1)	Long run $(t-1; t+n)$
	OLS 2nd	OLS 2nd	OLS 2nd	OLS 2nd
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)
Policy interventions				
Guarantees short run	7.860** (3.306)		7.709*** (2.254)	
Recapitalizations short run	3.539 (5.586)		2.779 (3.029)	
Liquidity injections short run	-0.231 (0.664)		-0.387 (0.446)	
Guarantees long run		0.741 (2.502)		0.760 (1.504)
Recapitalizations long run		-5.776 (4.091)		-4.114* (2.301)
Liquidity injections long run		1.348*** (0.240)		1.302*** (0.171)
Identification				
Inverse Mills ratio	97.699** (34.760)	91.051** (31.647)	101.995** (37.697)	97.240** (35.862)
Bank characteristics	YES	YES	YES	YES
Market and macro controls	YES	YES	YES	YES
Bank FE	YES	NO	YES	NO
Country × Year FE	YES	YES	YES	YES
Cluster	Banks	Banks	Banks	Banks
Observations	540	540	648	648
Number of banks	19	19	23	23
No of countries	18	18	20	20
Within R-squared	0.579	0.575	0.588	0.580

*, ** and *** denote significance levels of 10%, 5% and 1%

Table 8 Robustness assessment: period. This table reports the estimation results of the following regression: SystemicRisk_{ij,t} = $\beta_0 + \beta_1 \times Policy$ interventions_{ij, event window} + $\beta_2 \times Policy$ interventions_{ij, event window} × Peri $od_t + \beta_3 \times Period_t + \beta_4 \times IMR_{ij,t-1} + \Phi \times Bank \ controls_{ij,t-1} + \Psi \times Market \ \& \ Macro \ controls_{j,t-1} + \varphi_i + v_{jt} + \varepsilon_{ij,t-1}$ Panel A presents the results for the interaction of policy interventions with the global financial crisis period (GFC crisis), and, Panel B for the interaction of policy interventions with the sovereign debt crisis period (SVG crisis). Models (1) and (3) assess the short-term impact of policy interventions received by bank i from government j (i.e., interventions take the same value from one quarter before the event to one quarter after the event). Models (2) and (4) assess the long-term impact of policy interventions (i.e., interventions take the same value from one quarter before the event to the quarter they unwind). We report the results for the benchmark models (i.e., Column (4) Table 5 for short run effects, and, respectively, Column (7) Table 5 for long run effects. Method used is OLS Fixed Effects. The estimations control for the sample selection bias by including the inverse Mills ratio generated by the 1st stage Probit model. The dependent variable is the MES. The coefficients for period (GFC crisis, SVG crisis), bank characteristics (beta, size, asset growth, leverage, credit risk, liquidity, rollover risk, profitability, gross loans share, net non-interest margin), and market and macro control variables (competition, capital regulatory index, sovereign debt, inflation, GDP growth) are suppressed for brevity. Explanatory variables are one period lagged. All models include an unreported constant, country x year effects and bank fixed effects. Variables are winsorized at the 1st and 99th percentiles. Their definitions are given in Table 1. Standard errors (S.E.) clustered at bank level are reported in brackets

	A. Global fir	nancial crisis	B. Sovereign	debt crisis
	Short run $(t-1; t+1)$	Long run $(t-1; t+n)$	Short run $(t-1; t+1)$	Long run $(t-1; t+n)$
	OLS 2nd	OLS 2nd	OLS 2nd	OLS 2nd
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)
Policy interventions				
Guarantees short run	6.496** (2.529)		9.179*** (2.853)	
Recapitalizations short run	0.327 (1.705)		4.757 (6.769)	
Liquidity injections short run	0.293 (0.425)		0.039 (0.609)	
Guarantees long run		1.389 (1.637)		-2.013 (1.788)
Recapitalizations long run		-2.921 (2.145)		-6.213*** (2.213)
Liquidity injections long run		0.927*** (0.250)		1.042*** (0.242)
Policy interventions × Period				
Guarantees × Period	3.898 (3.913)		-2.362 (3.694)	
Recapitalizations × Period	4.548 (7.485)		-4.969 (7.011)	
Liquidity injections × Period	-0.369 (0.778)		0.959 (0.639)	
Guarantees after event × Period		-9.361** (4.161)		3.998** (1.433)
Recapitalizations after event × Period		-4.950 (10.938)		3.334 (3.276)
Liquidity injections after event × Period		0.420 (0.805)		0.070 (0.403)

Table 8 (continued)

	A. Global fin	ancial crisis	B. Sovereign of	lebt crisis
	Short run $(t-1; t+1)$	Long run $(t-1; t+n)$	Short run $(t-1; t+1)$	Long run $(t-1; t+n)$
	OLS 2nd	OLS 2nd	OLS 2nd	OLS 2nd
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)
Identification				
Inverse Mills ratio	102.074** (36.601)	102.261*** (33.276)	102.468*** (36.384)	99.089** (35.815)
Period	YES	YES	YES	YES
Bank characteristics	YES	YES	YES	YES
Market and macro controls	YES	YES	YES	YES
Bank FE	YES	YES	YES	YES
Country × Year FE	YES	YES	YES	YES
Cluster	Banks	Banks	Banks	Banks
Observations	710	710	710	710
Number of banks	30	30	30	30
No of countries	22	22	22	22
Within R-squared	0.579	0.575	0.579	0.573

*, ** and *** denote significance levels of 10%, 5% and 1%

(Model 4). The difference between the findings for the GFC period and the European sovereign debt crisis period indicates that government credibility has an important role. It is possible that during the GFC, market participants had high expectations of the governments' capacity to guarantee banks' debt, which increased the valuation of banks by investors. In turn, during the European sovereign debt crisis, governments' reliability in guaranteeing banks' debt was affected. The fiscal capacity of some European states declined considerably during the period 2011–2012, leading to a downgrade in sovereign ratings and increasing skepticism among market participants.

Methodology We further check the robustness of our findings by including the lags of the dependent variable, replacing the sum of MES by its median value within a quarter, and using the Delta CoVaR as an alternative method for computing systemic risk. Additionally, we re-estimate the empirical specifications using several alternative strategies: propensity score matching, a placebo test, and the difference-in-differences approach. The results are presented in Table 9.

Previous literature suggests that systemic risk measures can be persistent. For example, López-Espinosa et al. (2012) found a high level of persistence of systemic risk for a sample of large international banks. To account for this dynamic, we start by including the lag of the dependent variable among the regressors. The results from Panel A indicate that the positive impact of state guarantees on systemic importance in the short run and of liquidity injections in the long run is maintained. Unreported results confirm that the lagged MES carries a positive and significant coefficient, indicating a dynamic character of the banks' contribution to systemic risk.

Table 9 Robustness assessmer able. In Panel B we re-estimat CoVaR as dependent variable. run a <i>placebo test</i> assuming th 2007. In Panel F we run a <i>diff</i> . Column (7) Table 5 for long r loans share, net non-interest m crisis) are suppressed for brevit ated by the 1^{st} stage <i>Probit mou</i> tiles. Their definitions are given	nt: methodol e the benchn In Panel D v iat the rescue <i>erence-in-difi</i> run effects). argin), and n yy. Explanatc del. All mode n in Table 1.	ogy. This tab nark specific we apply a <i>p</i> <i>erences</i> anal <i>ferences</i> anal market and m narket and m ory variables els include a Standard err	ole presents ations using ations using <i>ropensity sco</i> ere providec lysis. We rej ents for ban nacro contro are one peri are one peri orts (S.E.) cl	robustness a the median <i>ore matching</i> ort the resu k characteris (variables ((iod lagged. 7 1 constant, c. ustered at b	mathefacture and the search of the MES for the granulysis an wo years early lis for the bestices (beta, s; stics (beta, s; competition, the estimatic ountry × year and level are	or different me e dependent vv d use a sampl ier. We assess inchmark mod ize, asset grov capital regula ons control for ons control for reported in br	sthodologies.] ariable. In Par e obtained thr i their impact lels (i.e., Colu wth, leverage, tory index, so the sample se ank fixed effe ackets	In Panel A we nel C we re-esti ough nearest-n on systemic ris mn (4) Table 5 credit risk, liq vereign debt, ii vereign debt, ii election bias by crs. Variables a	include the limate the em imate the em eighbor strat k using a pla for short ru uidity, rolloo, GDI including th including th re winsorizee	lags of the c ppirical mode egy (N = 1). eebo sampla neffects, an ver risk, pro rer risk, proP growth, GIe inverse Md at the 1st a	lependent vari- els using Delta In Panel E we e from 2005 to d, respectively, fitability, gross FC crisis, SVG Ills ratio gener- nd 99 th percen-
	A. Includin MES	g lagged	B. Median I	MES	C. Alternativ variable: Co	ve dependent VaR	D. Propensity ing	Score Match-	E.Placebo te	sst	F. Difference- in-Differences
	Short run $(t-1; t+1)$	Long run (t-1; t+n)	Short run (t-1; t+1)	Long run $(t-1; t+n)$	Short run $(t-1; t+1)$	Long run (t-1; t+n)	Short run (t-1; t+1)	Long run $(t-1; t+n)$	Short run $(t-1; t+1)$	Long run $(t-1; t+n)$	After event
Method	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	DID 2 nd
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)	CoVaR (5)	CoVaR (6)	MES (7)	MES (8)	MES (9)	MES (10)	MES (11)
Policy interventions											
Guarantees short run	4.946*** (1.709)		0.533^{***} (0.168)		3.458*** (1.029)		7.477*** (2.220)		0.572 (1.760)		
Recapitalizations short run	2.006 (2.671)		0.236 (0.241)		0.898 (1.785)		3.078 (3.354)		1.687 (3.321)		
Liquidity injections short run	0.346 (0.441)		0.014 (0.038)		-0.016 (0.234)		-0.072 (0.553)		0.537* (0.264)		
Guarantees long run		0.425 (1.043)		0.032 (0.109)		0.407 (0.724)		0.318 (1.721)		-0.927 (1.244)	
Recapitalizations long run		-1.984 (1.564)		-0.338** (0.151)		-1.646 (1.096)		-4.569** (2.169)		3.046 (4.047)	
Liquidity injections long run		0.652*** (0.162)		0.084^{***} (0.014)		0.463*** (0.067)		1.090^{***} (0.185)		3.631 (4.602)	
Guarantees × Post-inter- vention											-1.309 (1.875)

Table 9 (continued)											
	A. Includin MES	ıg lagged	B. Median I	MES	C. Alternativ variable: Co	/e dependent VaR	D. Propensity ing	Score Match-	E.Placebo to	est	F. Difference- in-Differences
	Short run $(t-1; t+1)$	Long run (t-1; t+n)	Short run (t-1; t+1)	Long run $(t-1; t+n)$	Short run $(t-1; t+1)$	Long run (t-1; t+n)	Short run (t-1; t+1)	Long run $(t-1; t+n)$	Short run (t-1; t+1)	Long run (t-1; t+n)	After event
Method	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	oLS 2 nd	OLS 2 nd	oLS 2 nd	OLS 2 nd	DID 2 nd
Dependent variable	MES (1)	MES (2)	MES (3)	MES (4)	CoVaR (5)	CoVaR (6)	MES (7)	MES (8)	MES (9)	MES (10)	MES (11)
Recapitalizations × Post- intervention											-4.158 (2.747)
Liquidity injections × Post- intervention											0.693^{**} (0.320)
Identification											
Inverse Mills ratio	65.401* (32.823)	62.363* (30.698)	8.392*** (2.608)	8.027*** (2.406)	44.962*** (15.731)	43.450*** (14.360)	104.785*** (36.548)	101.138^{***} (34.081)	-69.009 (59.827)	-48.305 (49.879)	100.009 *** (35.138)
Lagged Y	YES	YES	NO	NO	NO	NO	NO	NO	NO	NO	NO
Bank characteristics	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Market and macro controls	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Bank FE	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Country × Year FE	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Cluster	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks
Observations	710	710	710	710	710	710	692	692	124	124	710
Number of banks	30	30	30	30	30	30	30	30	30	30	30
No of countries	22	22	22	22	22	22	22	22	22	22	22
R -squared	0.654	0.649	0.589	0.583	0.591	0.581	0.581	0.573	0.404	0.399	0.564
*, ** and *** denote significan	nce levels of	10%, 5% and	1%								

Next, we use an alternative method for computing the quarterly values of the systemic risk indicator, which are derived from weekly values. Instead of summing up the weekly MES values within a quarter, we use their median value. The results from Panel B confirm that the coefficients associated with the main variables of interest remain valid.

In Panel C, we employ the *Delta CoVaR* (*Conditional Value at Risk*), developed by Adrian and Brunnermeier (2016), as the dependent variable. In contrast with *MES*, *Delta CoVaR* assesses the contagion effects from a bank to the system. Each bank's contribution to systemic risk is determined as the *VaR* of the system (i.e., the maximum possible loss of the system) conditioned on the event that each bank is at its own *VaR* level (i.e., the maximum possible loss of a bank). Online Appendix 4.2 provides a detailed description of the estimations. We note that the positive link of systemic importance with state guarantees in the short run and with liquidity injections in the long run is maintained.

In Panel D, we provide the results of a *propensity score matching* analysis. We construct an artificial control group by matching each intervened bank with the nearest non-intervened bank from our sample with similar characteristics. First, we run a probit model that estimates the propensity scores of all banks using the bank-level characteristics employed in our main regressions (i.e., beta, size, asset growth, leverage, credit risk, liquidity, rollover risk, profitability, gross loans share, and net non-interest margin). Employing a nearestneighbor matching strategy, we match each intervened bank with a non-intervened bank with the closest propensity score. Second, we rerun our main regressions using the matched samples. We observe that the size and significance of the findings remain unaltered.

In Panel E, we assume that the rescue packages were provided to banks two years earlier and use *a placebo sample* from 2005 to 2007. The coefficients associated with the policy interventions become insignificant, thus confirming the robustness of the impact.

Finally, we employ a *difference-in-differences* approach in Panel F to assess the difference between rescued banks before intervention versus rescued banks after intervention. In comparison with the long-term analysis where we examine the effects of interventions over several event windows (i.e., from one to four years), this model allows us to consider all quarters available after the bailouts took place within our dataset (which can imply a longer period). Policy interventions take the same value (i.e., the total volume of each type of bailout reported to banks' total assets) from one quarter before the event to all quarters after the event available within our dataset. The results point to the same positive and significant association of liquidity injections with banks' contribution to systemic risk in the long run and no significant impact of the other intervention mechanisms (Column (11)).

Other assessments In unreported results, we conduct additional robustness exercises. First, we re-estimate the benchmark models (Columns (4) and (7) from Table 5) employing alternative variables for funding risk and profitability. We replace the liquidity ratio with the loans to deposits ratio (computed as net loans to total deposits and borrowings), the interbank liquidity ratio (interbank assets to interbank liabilities), and the return on assets ratio with the operating profit margin (operating profit to average total assets). Second, we change the level of clustering of the standard errors from the bank level to the bank and quarter level (two-way clustering). Third, instead of dividing the policy interventions received by bank i in quarter t by total assets of the bank in the same quarter, we compute their weight in total assets in the previous quarter before implementation (t-1). Fourth, we separately include the policy intervention variables in the empirical specifications. The (unreported) results show no important differences from the results of the benchmark regression specification, and the impact of the policy interventions on systemic risk in terms of sign, size and significance remains unaltered.

5.2 Restrictions, policy interventions and systemic risk

Thus far, we have estimated the impact of policy measures on the "average" bank. This section presents the impact of restrictions imposed by regulators on the relation between emergency rescue actions and systemic risk. The model specification is introduced in subsection 3.3.2. We consider the following constraints: supervisory board intrusions, management pay limitations, and capital payout bans. Table 10 Panel A shows the empirical estimates for the short-run models, while Panel B shows the long-run specifications.

The results suggest that all three types of regulatory restrictions mitigate the harmful effect of liquidity injections on systemic risk in the short run, as captured by the negative and strongly significant coefficients associated with the interaction between restrictions and liquidity injections. These findings support the hypothesis that investors believe that liquidity injections enhance value when regulators impose tighter restrictions, leading to an increase in stock prices and therefore a reduction in systemic risk. It is likely that regulatory restrictions are perceived as effective tools in reducing portfolio risk by assuring stricter monitoring of the banks injected with liquidity. For the other type of financial assistance programs, the results show no significant effect of the interaction coefficient between bailouts and restrictions. A possible explanation is that in the short run, investors give greater importance to restrictions when they are associated with rescue packages that quickly fix banks' distress than they give to other interventions that may need a longer time horizon to produce their effect.

In the long run, we obtain a similar effect for supervisory board intrusions in the case of state guarantees and management pay limitations associated with recapitalizations. Seats on the supervisory board ensure stricter supervision of investment and lending practices after governments guarantee the debt of the financial institutions. This may counterbalance the lack of credibility regarding the governments' ability to guarantee banks' debt, softening the positive impact of state guarantees on systemic risk. In the case of recapitalizations, ceilings on executive salaries and bonuses may temper managers' appetite for risky projects, leading to an increase in market valuation. Therefore, executive compensation limits could further enhance the long-run beneficial impact of recapitalizations on systemic risk. We find no significant effect for liquidity bailouts, which indicates that the realization of distress continued in the long run at banks injected with liquidity, regardless of the type of regulatory restrictions applied.

Overall, applying restrictions to rescued banks while interventions are in effect can be an efficient policy tool to reduce the positive association of interventions with systemic risk. One limitation of our framework is that the interventions that we assess are specific to the global financial crisis and the sovereign debt crisis that affected European banks. Since then, a number of regulatory and legal changes have been implemented. For example, there have been several decisions by the European Commission that imposed stricter restrictions on the rescued banks, such as dividend bans, during the duration of the bailout.³⁷ Additionally, the overall legal framework changed after the adoption of the EU Bank Recovery and Resolution Directive (BRDD) in 2014.³⁸ However, governments can still choose from the bailout methods that we examined to save financial institutions in distress. In future research, it will be interesting to assess the effects of policy interventions on systemic risk considering the stricter restrictions imposed on rescued banks after 2014, especially in the context of the COVID-19 pandemic.

³⁷ The 2013 Banking Communication of the European Commission requires banks to be subject to dividend bans after receiving state aid in the form of recapitalization (European Commission 2013).

³⁸ Since January 1st, 2016, under the BRDD, banks are required to convert debt-like instruments to equity for a minimum amount of 8% (i.e., implement a bail-in) before receiving a bailout (European Commission 2014). This might shape the incentives of shareholders and creditors to monitor the risk of the financial institutions (Martino 2020).

Table 10 Policy interventio <i>icy interventions</i> _{<i>i</i>, <i>event wim</i> <i>controls</i>_{<i>i</i>,<i>i</i>,<i>i</i> + μ_{μ} + $e_{ij,r}$ Meth the inverse Mills ratio gener the inverse Mills ratio gener the inverse Mills ratio gener the event to one quar perfore the event to one quar performed as the event to one quarter to one quar performed as the event to one quarter to one quarter to one quarter to one quarter to one quarter performed as the event to one quarter to one quarter performed as the event performed as the event performed as the event to one quarter to one quarter to one quarter to one quarter to one quarter performed as the event performed as the even}}	ns by banks. I $h_{bw} + \beta_2 \times Policy.$ add used is OLS add by the \mathbb{R}^4 for the short-ti for the short-ti ter after the eve the output from on value from on value from on value from on the run effects, a llover risk, prof d effects. Varia	nteractions with restric <i>i interventions</i> _{ij, event} <i>trixed Effects.</i> The sam stage <i>Probit model.</i> The stage <i>Probit model.</i> The erm impact of policy ii ant) interacted with restrict ithe long-term impact of e quarter before the evolution if ability, gross loans sh C crisis, SVG crisis) arc bles are winsorized at	tions. This table re $_{indow} \times Restrictions_{ij}$ ple is restricted to 30 e dependent variable netrventions received interventions received i	ports the estimat $\mu_{I} + \beta_{3} \times Restrictions of the set of th$	on results or $2ns_{ij,i,j} + \beta_4 \times 2ns_{ij,i,j} + \beta_4 \times 2ns_{ij,i,j} + \beta_4 \times 2ns_{ij,i,j} + 2ns_{ij,i,j} \times 2ns_{ij,i,j} \times 2ns_{ij} \times 2ns_{ij}$	f the following regres $IMR_{y_{1}\ell_{1}} + \phi \times Bank$ donus and control for th flects the systemic ris i (i.e., interventions ta board intrusions, man es (i.e., interventions, man utlts for the models with the for bank characteri to control variables (c one period lagged. Al ven in Table 1. Stand	ssion: <i>SystemicRisk</i> , controlls _{<i>i</i>₁,1} + $\Psi \times M$ is sample selection i k of bank i from co ake the same value agenent pay limita received by bank i 1 irith country × year f stics (beta, size, as ompetition, capital II models include an ard errors clustered	$_{ii} = \beta_0 + \beta_j \times Pol-arket \& Macrooias by includinguntry j in quarterfrom one quarterfrom one quarterfrom governmenticon governmentixed effects (i.e.,et growth, lever-cegulatory index,unreported con-at bank level are$
	Panel A. Sh	nort run (t-1; t+1)			Panel B. Lo	ng run (t-1; t+n)		
	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd
Dependent variable	MES	MES	MES	MES	MES	MES	MES	MES
Restriction measure	Main results	Supervisory board intrusions	Management pay limitations	Capital payout bans	Main results	Supervisory board intrusions	Management pay limitations	Capital payout bans
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)
Policy interventions								
Guarantees	7.534*** (2.211)	7.596*** (2.580)	5.952** (2.564)	4.896* (2.592)	0.567 (1.508)	1.987 (1.291)	0.609 (1.207)	0.718 (1.343)
Recapitalizations	2.825 (3.216)	0.062 (3.814)	-5.215 (6.801)	-8.422	-4.248* (2.148)	-1.423 (1.919)	3.905 (3.740)	14.930 (11.219)
Liquidity injections	0.107 (0.559)	3.836** (1.926)	3.977* (2.036)	4.014** (2.029)	1.084^{***} (0.182)	1.757 *** (0.590)	(0.480)	0.861 (0.824)
Restriction								
Restriction		-9.557* (5.654)	-14.151*** (4.491)	-14.412*** (4.399)		0.713 (4.449)	-9.303*	-10.076^{*}
		((()	()

Table 10 (continued)								
	Panel A. Sho	rt run (t-1; t+1)			Panel B. Loi	ng run (t-1; t+n)		
	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd
Dependent variable	MES	MES	MES	MES	MES	MES	MES	MES
Restriction measure	Main results	Supervisory board intrusions	Management pay limitations	Capital payout bans	Main results	Supervisory board intrusions	Management pay limitations	Capital payout bans
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Interventions × Restriction								
Guarantees × Restriction		-0.613	5.171	5.745		-3.472^{***}	-0.446	0.185
		(3.253)	(4.646)	(4.118)		(1.329)	(1.606)	(1.713)
Recapitaliza-		4.848	8.430	10.125		-0.373	-7.939**	-18.350
tions × Restriction		(7.125)	(8.037)	(9.581)		(2.544)	(3.790)	(11.451)
Liquidity injec-		-4.212^{**}	-4.393^{**}	-4.436**		-0.775	-0.578	0.052
tions × Restriction		(1.927)	(2.024)	(2.012)		(0.582)	(0.487)	(0.790)
Identification								
Inverse Mills ratio	102.529 * * *	98.237***	89.222***	92.610^{***}	98.141^{***}	89.048***	95.000^{***}	88.285***
	(36.462)	(32.710)	(30.191)	(30.280)	(34.015)	(33.316)	(31.193)	(31.971)
Bank characteristics	YES	YES	YES	YES	YES	YES	YES	YES
Market and macro controls	YES	YES	YES	YES	YES	YES	YES	YES
Country × Year FE	YES	YES	YES	YES	YES	YES	YES	YES
Cluster	Banks	Banks	Banks	Banks	Banks	Banks	Banks	Banks
Observations	710	710	710	710	710	710	710	710
Number of banks	30	30	30	30	30	30	30	30
No of countries	22	22	22	22	22	22	22	22
Within R-squared	0.578	0.579	0.582	0.581	0.570	0.567	0.564	0.567

D Springer

 $^{*}, \, ^{**}$ and *** denote significance levels of 10%, 5% and 1%

5.3 Risk profiles, policy interventions and systemic risk

Finally, we examine the impact of banks' risk profiles on the relation between emergency rescue actions and systemic risk. The model specification is introduced in subsection 3.3.2. We discuss the empirical results for the following risk profile indices: size, leverage, and profitability. Table 11 Panel A shows the empirical estimates for the short-run estimates, while Panel B shows those for the long-run specifications.

Regarding size, the findings presented in Panel A Column (2) show that in the short run, the influence of recapitalizations is significantly different (although only marginally so) for larger banks in comparison to smaller banks, as suggested by the coefficient on the interaction term *Recapitalizations* × *Dummy Size* (i.e., 16.117*). For the average bank, the positive link of recapitalizations with systemic risk becomes significant, and it is increasing in size for larger banks. A possible explanation is that the market valuation of large banks injected with capital is penalized more because of the dilution of preintervention shareholders' earnings, which translates into an increase in systemic risk. In the long run, the positive association of liquidity injections with systemic risk is even more pronounced for large banks (Panel B Column (5)). The coefficient on the interaction term Liquidity injections × Dummy Size is positive and significant (i.e., 1.225**). This result is consistent with the literature on moral hazard embedded in government support programs for TBTF banks. Large banks are usually more focused on investment activities that are riskier than traditional lending activities; thus, rescue packages may incentivize them to increase portfolio risk, which enhances their systemic importance. In terms of policy implications, these findings support the restrictions suggested by the European Commission, which requested the downsizing of several large European banks that received bailouts during the crisis.

The link between interventions and systemic risk is also related to leverage in both the short and long run. The estimates show that the immediate influence of recapitalizations is negative for better capitalized banks (Column (3)). The interaction of *Dummy leverage* with recapitalizations enters the specifications with a negative sign (i.e., -19.833***).³⁹ The finding suggests that the contribution to systemic risk of banks injected with capital is lower immediately after the intervention for better capitalized banks than for less capitalized banks. The long-run results highlight that the realization of distress continues in the long run for banks injected with liquidity but is diminished when banks have a higher level of capitalization (Column (7)). These findings suggest that the provision of liquidity assistance should be oriented toward safer banks. This is in line with the actions of European governments, which provided liquidity assistance to safer financial institutions with regulatory capital above the minimum required threshold. They did not wait to intervene until bank capitalization deteriorated significantly, as it would have been costlier.

Finally, performance can be significantly associated with the relationship between recapitalizations and systemic importance in the long run, as shown by the interaction term $Recapitalizations \times Dummy$ Profitability in Column (8), which is positive and highly significant (i.e., 8.801***). The result indicates that recapitalizations fix the problems created by risk realizations, and the result is amplified for less profitable banks. In turn, for banks with higher performance, recapitalizations are linked with an increase in systemic risk. A possible explanation for this finding could be that more profitable banks are likely to borrow

³⁹ In unreported exercises, we interact a dummy that reflects whether a bank has Tier 1 ratio higher than the median regulatory capital ratio of the sample with bank-level interventions. The results yield similar conclusions.

Table 11 Policy interventions by bank: <i>icy interventions</i> _{<i>j</i>_i, <i>event window</i> +$\beta_2 \times R^c$ <i>controls</i>_{<i>j</i>_{i+1} + $\mu_{it} + \varepsilon_{ij}$. Method used is <i>O</i> the inverse Mills ratio generated by the t. Panel A shows the output for the shou before the event to one quarter after the term impact of rescue measures interact the event to the quarter they unwind). A Column (6) Table 5 for long run effect loans share, net non-interest margin), ar crisis) are suppressed for brevity. In the Leverage variable. In the model with d umeported constant and country × year bank level are reported in brackets}}	s. Interactions wit olicy intervention. US Fixed Effects. T I st stage Probit mot reterm impact of j event) interacted v ted with bank risk, ted with bank risk, ted with bank risk ted with and mac e model with dumi ummy Profitability fixed effects. Varia	In bank risk. Th s_{ji} , even window X, The sample is res deft. The depend policy interventic policy interventic its for the model is for bank chara for bank chara for outrol variak my Size, we exc s_{ij} , we exclude th bles are winsorii	is table reports Bank risk _{iji-1} +1 tricted to 30 ban ent variable is re- ans received by 1 offles (dummy Si offles (dummy Si ons received by 1 with country X- teteristics (beta, bles (competition lude the continuous RO ced at the 1 st and	the estimation re: $\beta_3 \times Bank$ risk $\mu_{ij,ri}$ ks that received in presented by MES pank i from gover rize, dummy Lever rize, dummy Lever rize, atset growth vear fit from gover the effects size, asset growth capital regulator ous Size variable. Exp 99 th percentiles. T	sults of the follov + $\beta_4 \times IMR_{g_{1,r_1}} + 9$ terventions and cc terventions and cc and reflects the sum nment j (i.e., inten age, dumm (3) tie., Column (3) , leverage, credit in the model with lanatory variables l'heir definitions a	wing regression: $b \times Bank \ control for the satisfies on the satisfies of the satisfies o$: SystemicRisk _{iji} , $sls_{ij,i} + \Psi \times Mari$ mple selection bit bank i from coun ne same value fro a shows the outp 3 shows the outp 3 shows the outp 3 shows the outp of shows the outp a spect, proi GDP growth, Gl age, we exclude age, we exclude e 1. Standard err	$= \beta_0 + \beta_1 \times Pol-$ zet & Macro us by including try j in quarter on one quarter of the long- quarter before quarter before quarter before quarter before crisis, SVG the continuous lels include an ors clustered at
	Panel A. Short I	run (t-1; t+1)			Panel B. Long	run (t-1; t+n)		
	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd	OLS 2 nd
Dependent variable	MES	MES	MES	MES	MES	MES	MES	MES
Bank risk	Main results	Size	Leverage	Profitability	Main results	Size	Leverage	Profitability
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)
Policy interventions								
Guarantees	7.534***	6.312*** (2.339)	9.358*** (2.992)	6.980** (2.978)	0.567 (1.508)	1.517 (1.838)	1.216	1.194
Recapitalizations	2.825	-1.633	7.987*	2.820	-4.248*	-2.411	-3.645**	-6.340***
	(3.216)	(3.090)	(4.227)	(3.972)	(2.148)	(2.343)	(1.824)	(1.967)
Liquidity injections	0.107 (0.559)	1.224 (1.009)	0.058 (0.506)	0.096 (0.539)	1.084^{***} (0.182)	0.074 (0.552)	1.217*** (0.222)	1.056^{***} (0.238)
Bank risk								

-8.451* (4.794)

3.667 (4.635)

0.995 (16.656)

-2.800 (4.405)

6.899* (3.768)

-1.408 (11.871)

Bank risk_{t-1}

Table 11 (continued)								
	Panel A. Short	run (t-1; t+1)			Panel B. Long	run (t-1; t+n)		
	OLS 2 nd							
Dependent variable	MES							
Bank risk	Main results	Size	Leverage	Profitability	Main results	Size	Leverage	Profitability
	(1)	(2)	(3)	(4)	(2)	(9)	(1)	(8)
Interventions $ imes$ Bank risk								
Guarantees \times <i>Bank risk</i> _{<i>i</i>-1}		8.784	-1.696	1.374		-1.540	0.168	1.250
		(6.061)	(4.104)	(3.619)		(2.154)	(1.879)	(0.938)
Recapitalizations × Bank risk _{r-1}		16.117*	-19.833^{***}	2.438		-2.178	2.110	8.801^{***}
		(8.870)	(5.531)	(5.047)		(3.656)	(2.205)	(2.367)
Liquidity injections $\times Bank risk_{i-1}$		-1.448	0.635	0.359		1.225^{**}	-0.939*	0.029
		(1.118)	(1.246)	(0.761)		(0.576)	(0.486)	(0.256)
Identification								
Inverse Mills ratio	102.529 ***	98.539***	95.307***	101.051^{***}	98.141***	90.900***	91.921***	93.212***
	(36.462)	(29.884)	(31.811)	(31.157)	(34.015)	(28.997)	(29.313)	(28.130)
Bank characteristics	YES							
Market and macro controls	YES							
Country × Year FE	YES							
Cluster	Banks							
Observations	710	710	710	710	710	710	710	710
Number of banks	30	30	30	30	30	30	30	30
No of countries	22	22	22	22	22	22	22	22
Within R-squared	0.578	0.587	0.583	0.575	0.570	0.567	0.567	0.571
*, ** and *** denote significance levels	of 10%, 5% and 1	%						

202

more and engage in risky operations, increase their portfolio risk, and therefore intensify their systemic importance.

Overall, the results suggest that the relation between policy interventions and systemic contribution varies significantly with the risk profiles of banks. Characteristics such as size, leverage, and profitability can significantly shape the relationship between bailouts and banks' systemic importance in the long run, while the immediate link between governmental assistance programs and systemic risk is heterogeneous among banks with different sizes and levels of leverage. Most importantly, from supervisors' perspective, the efficiency of emergency rescue measures can be mitigated or enhanced by banks' risk strategies.

6 Conclusion

In this paper, we investigate how policy interventions are associated with banks' contribution to systemic risk in the short and long run. Using a unique bank-level dataset that consists of 83 banking institutions from 22 European countries, we estimate systemic risk based on the loss generated by the reduction in banks' market capitalization under extreme events, employing Acharya et al.'s *MES* (2017), a measure that reflects both risk realizations and forward-looking risk. The estimations are performed for the 2005–2014 period.

Analyzing a large and original bank-level dataset on policy interventions, we then show that the bailouts are associated with different evolutions of systemic risk in the short and in the long run. We employ the Heckman selection approach and control for a variety of bank, market, and macro characteristics. In the first stage, we use a *Probit* model to estimate the probability of a bank experiencing an intervention. In the second stage, we examine the association of policy interventions with systemic risk using an *OLS fixed effects* model.

Our findings provide evidence that in the short run, banks that receive policy interventions are linked with an enhanced contribution to systemic risk, indicating a delayed effect of bailouts in fixing systemic distress. As interventions are usually applied after risk is realized, the declines in market valuation of rescued banks continue after the bailouts are implemented. The effect is strongly significant for state guarantees and might be explained by the credibility issues faced by European governments. In the long run, banks injected with liquidity remained weak, and investors penalized their stock returns, as reflected in a higher systemic contribution, while recapitalizations fixed the problem created by risk realizations.

We further provide empirical evidence that policy interventions relate differently with systemic risk across banks when regulatory restrictions are imposed and where bank risk strategies differ. In sum, the picture that arises is one in which the positive association of *guarantees* with systemic risk is weaker when the regulator appoints members to the supervisory board in the long run. The immediate influence of *recapitalizations* is negative for small and better capitalized banks, while in the long run, recapitalizations are associated with reduced systemic importance, especially for less profitable banks and in cases in which the regulator limits management pay. In the short run, injecting liquidity can reduce systemic risk when the regulator imposes restrictions such as supervisory board intrusions, management pay limitations, and capital payout bans. In the long run, the positive link between *liquidity injections* and systemic risk is mitigated for small or better capitalized banks.

Our findings suggest that banks' regulatory restrictions and risk profiles should play a key role in the design of optimal financial assistance programs. The effectiveness of policy interventions can be significantly altered by regulatory constraints and banks' risk strategies.

Supplementary Information The online version contains supplementary material available at https://doi. org/10.1007/s10693-023-00404-8.

Acknowledgements We thank Mark Carey (the editor) for his extraordinary generosity in providing many comments that greatly helped improve our manuscript and two anonymous referees for many most helpful comments. Nistor acknowledges financial support from a grant of the Ministry of Research, Innovation and Digitization, CNCS/CCCDI – UEFISCDI, project number PN-III-P1-1.1-TE-2019-1522, within PNCDI III. Ongena acknowledges past financial support from ERC ADG 2016—GA 740272 lending.

Funding Open access funding provided by University of Zurich

Data Availability The raw data required to reproduce the above findings are available upon request.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Acharya VV, Yorulmazer T (2007) Too many to fail an analysis of time inconsistency in bank closure policies. J Financ Intermed 16(1):1–31
- Acharya V, Yorulmazer T (2008) Cash-in-the-market pricing and optimal resolution of bank failures. Rev Financ Stud 21:2705–2742
- Acharya VV, Engle RF, Richardson M (2012) Capital Shortfall: A new approach to ranking and regulating systemic risks. Am Econ Rev 102(3):59–64
- Acharya V, Pedersen L, Philippon T, Richardson M (2017) Measuring systemic risk. Rev Financ Stud 30(1):2–47
- Adrian T, Brunnermeier MK (2016) CoVaR. Am Econ Rev 106(7):1705-1741
- Allen F, Carletti E, Goldstein I, Leonello A (2015) Moral hazard and government guarantees in the banking industry. J Financ Regul 1(1):30–50
- Anginer D, Demirgüç-Kunt A, Zhu M (2014a) How does competition affect bank systemic risk? J Financ Intermed 23(1):1–26
- Anginer D, Demirgüç-Kunt A, Zhu M (2014b) How does deposit insurance affect bank risk? Evidence from the recent crisis. J Bank Finance 48:312–321
- Barth JR, Caprio G, Levine R (2013) Bank regulation and supervision in 180 countries from 1999 to 2011. J Financ Econ Policy 5(2):111–219
- Bayazitova D, Shivdasani A (2012) Assessing TARP. Rev Finan Stud 25(2):377-407
- Behn B, Haselmann R, Kick T, Vig V (2017) The political economy of bank bailouts. Working paper. Available online at https://finance.unibocconi.eu/sites/default/files/files/media/attachments/paper_ v0815_Bocconi20170906145836.pdf
- Berger AN, Roman RA (2015) Did TARP banks get competitive advantages? J Financ Quant Anal 50:1199–1236
- Berger AN, Roman RA, Sedunov J (2020) Do bank bailouts reduce or increase systemic risk? The effects of TARP on financial system stability. J Financ Intermed 43:100810
- Berger AN, Nistor S, Ongena S, Tsyplakov S (2022) Catch, restrict, and release: the real story of bank bailouts. Swiss Finance Institute Research Paper, No. 20–45, Swiss Finance Institute
- Black L, Correa R, Huang X, Zhou H (2016) The systemic risk of European banks during the financial and sovereign debt crises. J Bank Finance 63:107–125
- Bostandzic D, Pelster M, Weiß GNF (2014) Systemic risk, bank capital, and deposit insurance around the world. Working Paper. Available at SSRN: https://ssrn.com/abstract=2438693
- Braun M, Raddatz C (2010) Banking on politics: When former high-ranking politicians become bank directors. World Bank Econ Rev 24(2):234–279

- Brei M, Gambacorta L, von Peter G (2013) Rescue packages and bank lending. J Bank Finance 37(2):490–505
- Brei M, Gadanecz B (2012) Public recapitalisations and bank risk: Evidence from loan spreads and leverage. BIS Working Papers, No. 383
- Brown CO, Dinc IS (2005) The politics of bank failures: Evidence from emerging markets. Quart J Econ 120:1413–1444
- Brown CO, Dinç IS (2011) Too many to fail? Evidence of regulatory forbearance when the banking sector is weak. Rev Finan Stud 24(4):1378–1405
- Brownlees C, Engle R (2017) SRISK: A Conditional Capital Shortfall measure of systemic risk. Rev Financ Stud 30(1):48–79
- Brunnermeier MK, Oehmke M (2013) Bubbles, financial crises, and systemic risk. Handb Econ Finan 2:1221–1288
- Brunnermeier MK, Dong GN, Palia D (2020) Banks' noninterest income and systemic risk. Rev Corp Finan Stud 9(2):229–255
- Buch CM, Krause T, Tonzer L (2019) Drivers of systemic risk: Do national and European perspectives differ? J Int Money Financ 91:160–176
- Cao J, Illing G (2008) Liquidity shortages and monetary policy. CESifo Working Paper Series 2210, CESifo Group, Munich
- Choi DB (2014) Heterogeneity and stability: Bolster the strong, not the weak. Rev Financ Stud 27(6):1830–1867
- Cordella T, Yeyati E (2003) Bank bailouts: Moral hazard vs. value effect. J Financ Intermed 12(4):300-330
- Cruz C, Keefer P, Scartascini C (2016) Database of Political Institutions Codebook, 2015 Update (DPI2015). Inter-American Development Bank. Updated version of Thorsten Beck, George Clarke, Alberto Groff, Philip Keefer, and Patrick Walsh, 2001. New tools in comparative political economy: The Database of Political Institutions." 15:1, 165–176 (September), World Bank Economic Review
- Dam L, Koetter M (2012) Bank bailouts and moral hazard: evidence from Germany. Rev Finan Stud 25(8):2343-2380
- De Santis RA (2014) The euro area sovereign debt crisis: identifying flight-to-liquidity and the spillover mechanisms. J Empir Finan 26:150–170
- Demirgüç-Kunt A, Detragiache E (2002) Does deposit insurance increase banking system stability? An empirical investigation. J Monet Econ 49(7):1373–1406
- Demirgüç-Kunt A, Huizinga H (2004) Market discipline and deposit insurance. J Monet Econ 51(2):375-399
- Demirgüç-Kunt A, Huizinga H (2010) Bank activity and funding strategies: The impact on risk and returns. J Financ Econ 98(3):626–650
- Dowd K (1998) Beyond Value-at-Risk: The new science of risk management. John Wiley & Sons, London
- Duchin R, Sosyura D (2014) Safer ratios, riskier portfolios: Banks' response to government aid. J Financ Econ 113(1):1–28
- European Commission (2009) Report on competition policy. Available online at: http://ec.europa.eu/competition/publications/annual_report/2009/en.pdf. Accessed Feb 2023
- European Commission (2013) Communication from the Commission on the application, from 1 August 2013, of State aid rules to support measures in favour of banks in the context of the financial crisis ('Banking Communication'). Official Journal of the European Union, 2013/C 216/01
- European Commission (2014) Directive 2014/59/EU of the European Parliament and of the Council of 15 May 2014 establishing a framework for the recovery and resolution of credit institutions and investment firms. Official Journal of the European Union, L 173/190
- European Stability Mechanism (2013) Annual Report. Available online at: https://www.esm.europa.eu/publi cations/esm-annual-report-2013. Accessed Feb 2023
- Farhi E, Tirole J (2012) Collective moral hazard, maturity mismatch and systemic bailouts. Am Econ Rev 102(1):60–93
- Flannery MJ, Hankins KW (2013) Estimating dynamic panel models in corporate finance. J Corp Finan 19:1–19
- Freixas X, Giannini C, Hoggarth G, Soussa F (1999) Lender of last resort: A review of the literature. In: Goodhart C, Illing G (eds) Financial Crises, Contagion, and the Lender of Last Resort. A Reader. Oxford University Press 2002
- Gerhardt M, Vander Vennet R (2017) Bank bailouts in Europe and bank performance. Financ Res Lett 22:74-80
- Giannetti M, Simonov A (2013) On the real effects of bank bailouts: Micro evidence from Japan. Am Econ J Macroecon 5(1):135–167

- Girardi G, Ergün AT (2013) Systemic risk measurement: Multivariate GARCH estimation of CoVaR. J Bank Finance 37(8):3169–3180
- Gropp R, Gruendl C, Guettler A (2014) The impact of public guarantees on bank risk taking: Evidence from a natural experiment. Rev Financ 18(2):457–488
- Heckman JJ (1979) Sample selection bias as a specification error. Econ J Econ Soc 47(1):153-161
- Homar T (2016) Bank recapitalizations and lending: a little is not enough. ERSB Working Paper Series No. 16/2016
- Honohan P, Klingebiel D (2003) The Fiscal Cost Implications of an accommodating approach to banking crises. J Bank Finance 21:1539–1560
- Huang R, Ratnovski L (2009) Why Are Canadian banks more resilient? IMF Working Paper 09/152
- Jorion P (1997) Value-at-Risk: The new benchmark for controlling market risk. Irwin, Chicago
- Kane EJ (1995) Three paradigms for the role of capitalization requirements in insured financial institutions. J Bank Finance 19(3–4):431–459
- Karamysheva M, Seregina E (2020) Prudential policies and systemic risk: The role of interconnections. NRU Higher School of Economics, Moscow RU
- Kick T, Koetter M, Poghosyan T (2016) Bank recapitalization, regulatory intervention, and repayment. J Money Credit Bank 48(7):1467–1494
- Koetter M, Noth F (2016) Did TARP distort competition among sound unsupported banks? Econ Inq 54(2):994–1020
- Laeven L, Valencia F (2012) The use of blanket guarantees in banking crises. J Int Money Financ 31:1220–1248
- Laeven L, Ratnovski R, Tong H (2016) Bank size and systemic risk: Some international evidence. J Bank Finance 69(1):S25–S34
- Liu W-M, Ngo P (2014) Elections, political competition and bank failure. J Financ Econ 112:251-268
- López-Espinosa G, Moreno A, Rubia A, Valderrama L (2012) Short-term wholesale funding and systemic risk: A global *CoVaR* approach. J Bank Finance 36(12):3150–3162
- Martino E (2020) The bail-in beyond unpredictability: Creditors' incentives and market discipline. European Business Organization Law Review
- Martynova N, Ratnovski L, Vlahu R (2020) Bank profitability, leverage constraints, and risk-taking. J Financ Intermed 44:100821
- Mayordomo S, Rodriguez-Moreno M, Peña JI (2014) Derivatives holdings and systemic risk in the U.S. banking sector. J Bank Finance 45:84–104
- Mishkin FS (2006) How big a problem is too big to fail? A review of Gary Stern and Ron Feldman's too big to fail: The hazards of bank bailouts. J Econ Lit 44(4):988–1004
- Myerson RB (2012) A model of moral-hazard credit cycles. J Polit Econ 120(5):847-878
- Nickell S (1981) Biases in dynamic models with fixed effects. Econometrica 49:1417-1426
- Panetta F, Faeh T, Grande G, Ho C, King M, Levy A, Signoretti FM, Taboga M, Zaghini A (2009) An assessment of financial sector rescue programmes. Bank of Italy, Economic Research and International Relations Area, No. 47
- Perotti E, Ratnovski L, Vlahu R (2011) Capital regulation and tail risk. Int J Cent Bank 7(4):123-163
- Petrovic A, Tutsch R (2009) National rescue measures in response to the current financial crisis. ECB Legal Working Paper Series, No 8/July 2009
- Repullo R (2005) Liquidity, risk taking, and the lender of last resort. Int J Cent Bank 1(2):47-80
- Saunders A (1999) Financial Institutions Management: A modern perspective, 3rd edn. Irwin Series in Finance, McGraw-Hill, New York
- Stock JH, Yogo M (2005) Testing for weak instruments in linear IV regression. NBER Technical Working Paper 284. http://www.nber.org/papers/T0284. Accessed Feb 2023
- Tarashev N, Borio C, Tsatsaronis K (2010) Attributing systemic risk to individual institutions. BIS Working Papers No. 308
- Weiß GNF, Bostandzic D, Neumann S (2014a) What factors drive systemic risk during international financial crises? J Bank Finance 41:78–96
- Weiß GNF, Neumann S, Bostandzic D (2014b) Systemic risk and bank consolidation: International evidence. J Bank Finance 40:165–181

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.